# How Effective is the Education Response for Displaced Syrian Children in Lebanon?

Non-Formal Education Programming in a Fragile Context.

BACHELOR THESIS PUBLIC GOVERNANCE ACROSS BORDERS ALINA KLEHR

July 4th, 2018

University of Twente

5. Drienerlolaan. 7522 NB Enschede

The Netherlands

## Abstract

The Bachelor thesis addresses the research question "how and to what extent do selected education projects undertaken by non-state service providers address the factors inhibiting the access to education for Syrian refugees in Lebanon effectively?". Drawing on service delivery theory in fragile states in the education sector, the study presents the main concepts education delivery, state fragility, education in emergencies, effectiveness, and barriers to education. Deploying a longitudinal case study approach, the thesis takes four education projects derived from qualitative desk research into account. By analysing the three sub-research questions (1) "What are the factors that constrain access to education for Syrian refugees in Lebanon?", (2)"What are the current initiatives taken by national and international NGOs regarding supporting education among Syrian refugees in Lebanon?", and (3) "What is the documented effectiveness of the selected education projects in reducing the main barriers to access to education for Syrian refugees?", the thesis consequently arrives at the conclusion that the education delivery is constrained by the education barriers cost to education, negative coping mechanisms, psychological stress and trauma, being over age, lacking language skills, lacking documentation, and discrimination and violence. It identifies that non-state education delivery significantly contributes to the provision of education for marginalized groups. However, structural barriers such as capacity gaps and fragmentation of service delivery require to be addressed beyond the NGO and INGO interventions in order to create sustainable improvement.

## Table of Content

Abstract	ii
1. Background: Introduction and Research Question	1
2. Theory	4
2.1. New Public Management in a Stable Environment	4
2.1.1 Contracting Out	5
2.2 State Fragility	6
2.2.1 Service Delivery in Fragile Situations	7
2.3 Education Delivery in Fragile Situations and Education in Emergencies	9
2.3.1 Defining Non-State Education Delivery	10
2.3.2 Education Delivery Through NGO-Provision	12
2.3.3 Conceptualization of the Effectiveness of Education Delivery	13
2.3.4 Conceptualization of the Barriers to Education	14
3. Data and Documents	17
3.1 Methodology (Research Design, Data Analysis Method and Limitations)	17
3.2 Operationalization and Data Collection Methods	19
4. Analysis	21
4.1 What are the Factors that Constrain the Access to Education for Syrian Refugees in Lebanon	21
4.1.1 The Refugee Crisis and the Response in Lebanon	21
4.1.2 The Lebanese Socio-Economic Situation	23
4.1.3 The Lebanese Education Sector	24
4.1.4 The Adjustments in the Lebanese Education Sector	25
4.1.5 Answer to the First Sub-Research Question: The Factors that Constrain the Acces Education for Syrian Refugees	
4.2 What are the Current Initiatives Taken by National and International NGOs Regarding Suppor	rting
Education Among Syrian Refugees in Lebanon?	32
4.2.1 An Overview of Education Activities by NGOs and INGOs	32
4.2.2 Answer to the Second Sub-Research Question: The Current Initiatives Taken by National International NGOs Regarding Supporting Education Among Syrian Refugees in Lebanon	
Project 1: The Back-to-School Campaign by MEHE, UNHCR, UNICEF and Partner NGOs	35

Project 2: Learning in a Healing Classroom Programme by the International Rescue Committee37
Project 3: Non-Formal Education Programming by the Norwegian Refugee Council
Project 4: The Rainbow of Hope Project by Search for Common Ground
4.3 What is the Documented Effectiveness of the Selected Education Projects in Reducing the Main Barriers to Access to Education for Syrian Refugees?
Project 1: The Back-to-School Campaign by MEHE, UNHCR, UNICEF and Partner NGOs 43
Project 2: Learning in a Healing Classroom Programme by the International Rescue Committee44
Project 3: Non-Formal Education Programming by the Norwegian Refugee Council 46
Project 4: The Rainbow of Hope Project by Search for Common Ground
5. Conclusion
6 References
7. Data Appendix
7.1 Historical Background of Lebanon: Conflict and Sectarian Tensions as Fragility Drivers 63
7.2 Operationalization Table

## 1. Background: Introduction and Research Question

This thesis addresses the issue that a high proportion of displaced Syrian youth in Lebanon lacks access to adequate education opportunities. While state authorities, international organizations and non-governmental organizations (NGOs) seek to overcome barriers to education with a wide range of projects, the projects' effectiveness in tackling these barriers remains in the dark.

Since the beginning of the Syrian war in 2011, Lebanon experienced the worldwide highest per capita influx of displaced persons with 1.5 million Syrians seeking refuge, which exceeds 30 percent of the country population (Ministry of Education and Higher Education [MEHE], 2014; United Nations System in Lebanon [UN Lebanon], 2017). Given the impact of confessionalism on the Lebanese political system that paralyses decision-making and makes it prone to external influence, as well as the negative economic repercussions of the 1975-1990 civil war and the 2006 conflict with Israel, Lebanon already experienced difficulties in the public sector prior to the arrival of the Syrian refugees (The World Bank Group Middle East and North Africa Region, 2015). The abrupt population growth and its accompanying changing composition not only disrupts Lebanon's confessional balance but further exacerbates the pressure on the state infrastructure that remains behind the population's need in terms of social service delivery.

Around 42 percent of the displaced Syrians are school-aged children between 3-18 years (MEHE, 2014). According to a study conducted by UNESCO, UNFPA, UNHCR, UNICEF, and Save the Children International (2014) only 6 percent of the surveyed Syrian refugee youth are enrolled in formal education in Lebanon, even though 35 percent of the youth sample have been previously enrolled in Syria. Whilst also catering for the Lebanese youth, the education demand exceeds the capacity and the resources of the Lebanese education sector. This leads to the problem that a large share of children and youth lack access to appropriate education opportunities, leaving many of them out of school.

Addressing this issue by guaranteeing the human right to education plays a vital role for the displaced Syrian children as well as the Lebanese host community. For the affected children, education in the immediate reaction to emergencies provides a protective environment that can serve as entry point for psychological support and other life-saving information e.g. on self-protection from abuse. It furthermore opens the opportunity for reaching children with other crucial services such as nutrition, health, or water and sanitation (Baird, 2010; Burde, Kapit, Wahl, Guven, & Skarpeteig, 2017). In conflict-situations, education also reduces the likelihood of being recruited for combat, forces labour, drug trafficking or sexual exploitation (Save the Children, 2015). In the long-term, education can offer a sense of normalcy within a disrupted life by providing familiar routines and a perspective for the future. It also sustains progress already made by school-going children and decreases the impact of

interruptions. Beyond the affected individual, education can contribute to a shared identity through communicating values such as human rights and tolerance, thereby fostering integration in the host-community. The acquired skills enable an economically independent life and increased future job opportunities which contain implications for the country's social capital and economic development (Baird, 2010; Burde et al., 2017). In situations of conflict and fragility, education contains an even higher relevance as it is prone to politicization. If addressed effectively, education delivery can assist in peace-building processes and promote social cohesion (Rose, 2007a). Next to the benefits that come along with increased education opportunities, paying attention to this issue is also relevant because states need to be responsive to their population's needs. According to 16 studies that were conducted across different types of emergencies by eight organizations, 99 percent of children in crisis situations see education as a priority (Save the Children, 2015). This is also the case in Lebanon. It is known that among the Syrian refugees that were enrolled in Syria, 87 percent would like to continue their education in Lebanon if possible (UNESCO et al., 2014).

Being aware of these wide ranging effects of lacking access to education, the Lebanese government adopted the Lebanon Crisis Response Plan (LCRP) in 2015 and launched the Reaching All Children with Education programme in 2014 as respective response for the education sector (Government of Lebanon [GoL] & United Nations [UN], 2015; MEHE, 2014). As the Lebanese state alone is unable to cater the needs of all Syrian refugee children, the response to the education crisis involves numerous non-state actors that support the cause. National and international NGOs (INGOs) as well as international organizations provide complementary educational programmes and prepare the refugee youth for the transition into the formal education system. Yet despite the efforts that are undertaken by all actors, barriers to education remain.

Therefore, the question arises to what extent the response to the education crisis by non-state actors addresses these barriers effectively. In the past decade, academic research on education in crisis situations and early reconstruction has been growing (Kagawa, 2005). This body of literature includes best practices for education delivery in fragile situations, improving access for marginalised and vulnerable communities, and the role of NGOs in doing so (Rose, 2007a; Rose & Greeley, 2006). What is missing according to a literature review on what works for the education delivery in fragile situations by Burde, Guven, Kelcey, Lahmann, and Al-Abbadi (2015) is an examination of the best interventions for supporting educational access, quality of learning and well-being for refugee children and youth. Therefore, Burde et al. (2015) suggest comparing the impact of different access interventions as number one recommendation for further research.

Taking the state of research in the field into account, this paper seeks to address this research gap by examining education projects carried out by different non-state actors in terms of their effectiveness in addressing the main barriers to education for Syrian refugee children and youth in Lebanon. The paper hereby considers projects carried out by different non-state actors such as NGOs, INGOs and international organizations. This paper adds to the existing body of knowledge by pointing out similarities and differences between the education access interventions and assessing their effectiveness in reducing or overcoming the barriers to education. The barriers are derived from various reports that are partly based on the input and feedback of the affected refugee youth. That way it can by guaranteed that the projects are demand-oriented and not exclusively formulated by the donor community. With this approach, insights into the informal education sector in Lebanon are provided and knowledge on the effectiveness, the paper takes a stance on the importance of ownership among the affected population, which possibly contributes to future project implementation processes in fragile contexts in the education sector.

Consequently, this paper raises the research question:

How and to what extent do selected education projects undertaken by non-state service providers address the factors inhibiting the access to education for Syrian refugees in Lebanon effectively?

This descriptive main-research question is structured in three sub-research questions.

#### (1) What are the factors that constrain access to education for Syrian refugees in Lebanon?

This sub-question is designed to introduce the specific country context of Lebanon. Besides explaining Lebanon's particular political-administrative system and historical background of civil war, it sheds light on the education sector before and after the influx of Syrian refugees. It finishes by pointing out the key factors that hinder Syrian refugees from starting or continuing their education.

(2) What are the current initiatives taken by national and international NGOs regarding supporting education among Syrian refugees in Lebanon?

This sub-question provides an overview on different NGO and INGO education projects. It continues by analysing four education projects in-depth regarding the extent to which they address the barriers to education identified in sub-research question one.

*(3)* What is the documented effectiveness of the selected education projects in reducing the main barriers to access to education for Syrian refugees?

This sub-question takes into account the extent to which the projects deploy monitoring mechanisms and the extent to which the project outputs correspond with the formulated project objectives.

The underlying assumption is that the refugees have a great awareness of their own educational needs and what prevents them from satisfying them, and that by orienting the project interventions towards this knowledge, the projects' effectiveness increases.

Following the introduction of the research problem and the presentation of the research question and its respective sub-research questions, chapter two outlines the theoretical background on education delivery in a fragile context. Chapter three describes the methodology and operationalization which explains how the data used in the analysis has been extracted and displays the data that is used to answer the research question and the sub-questions. Chapter four analyses and answers the three sub-research questions, after which chapter five provides the conclusion and reflection. The research questions as well as the sub-research questions are answered in the conclusion chapter which also provides recommendations for further research.

## 2. Theory

This chapter presents the theoretical grounding for the topic of limited access to education for refugees in a fragile state context. Section 2.1 discusses New Public Management as framework theory for service delivery in stable environments and particularly focuses on contracting out for social services in section 2.1.1. Section 2.2 elaborates on the fragile state context and section 2.2.1 describes its implications for service delivery. Section 2.3 zooms in and presents education delivery in fragile situations and introduces the concept of emergency education. Section 2.3.1 defines non-state education delivery, with a particular focus on NGO provision in section 2.3.2. Section 2.3.3. conceptualized the first core concept, effectiveness, and section 2.3.4 provides the conceptualization for the second core concept, the barriers to education.

### 2.1. New Public Management in a Stable Environment

A stable environment is in most cases a state, which consists according to Jellinek's three-element doctrine of a state territory, a state population and a state authority, which represents the state's ability to execute its core functions (Hirschmann, 2016). A stable state exercises four core functions: (1) security governance through its monopoly of power, (2) political governance through the rule of law, (3) administrative governance through the execution of legislative decisions, (4) and socio-economic governance through the provision of a minimum of welfare, social services and mechanisms for resource redistribution (Hirschmann, 2016). The *access to education* can be classified as a provision of a public service in the education sector.

The contemporary management theory New Public Management (NPM) can be deployed as framework concept for service delivery in stable environments. NPM is a "global public management

reform movement that has redefined the relationships between government and society" (Frederickson, Smith,, Larimer, & Licari, 2016, p. 229). The connections between government and society refer to the implementation of public policies and programmes which makes NPM a governance model of organisational and institutional reform. NPM revolves around six core issues: productivity, marketing, decentralization, policy, service orientation and accountability (Frederickson et al., 2016). Especially the last two dimensions are of importance for this paper. The aspect of service orientation refers to a stronger connection between governments and citizens which aims at improving the customer satisfaction of public services. Accountability refers to governments that deliver on their promises.

There is considerable academic debate on the relationship between the public and private sector and their role in providing public services. NPM holds the explanatory targets of improving the efficiency of the public sector and its responsiveness to citizen demands through the application of corporate values and practices. Furthermore, for NPM the public sector differs from the private sector only by the type of goods and services being produced, which reduces the government to "a contractual agent for [...] citizens" (Frederickson et al., 2016, p. 266). Therefore, contracting out for efficiency represents one of the core dimensions of NPM.

#### 2.1.1 Contracting Out

Contracting out is a form of public management in which public service delivery is not only provided by the state but also by privately-owned enterprises and non-profit organizations. Due to very different sector characteristics and bidding environments, contract management differs significantly among the three realms (1) capital construction and repair and maintenance purposes, (2) weapon systems, airplanes or space projects, (3) and the field of social services (Frederickson et al., 2016).

In the realm of social services, to which education belongs, contracts are usually not put out for bids but negotiated with one provider (Frederickson et al., 2016). Contracting out for social services is characterized by limited markets, hardly definable goals, and difficulties in measuring outcomes, which prevent the rise of market or government monopolies (Frederickson et al., 2016). Therefore, contracting out for social services can be understood as negotiated network in which the government sets out broad policy objectives (steering) and policy networks implement the policy through entrepreneurial activity (rowing) (Frederickson et al., 2016).

The shifted responsibilities raise questions on the extent to which governments can be held accountable, as contracting exports not only day-to-day government activities to third-party contractors such as NGOs, but also much of the capacity to direct and to control policy. By creating a supply and demand driven environment in which service providers compete for satisfying clientele demands, NPM seeks to circumvent this accountability issue but, in turn, creates regulatory capture with services providers being responsive to their clientele instead of the public interest (Frederickson et al., 2016).

Because of that, managing by contract requires effective contract management that sets out clear outcome measurements to determine effectiveness and that offers tools to hold the parties of the negotiated network accountable. In order to become smart and effective in buying, governments are advised to prevent contracting out for core government functions and to provide direct funding to stable, centralized, monopolistic service providers (Frederickson et al., 2016). In emergency situations, stable networks and cooperative relationships between the public and private sector are equally important (Frederickson et al., 2016).

However, Meier and O'Toole (2009) in Frederickson et al. (2016) criticise that "contracting out is often not done for reasons of performance and efficiency but rather to get rid of problems" (Frederickson et al., 2016, p. 114). They point out that networks can be effective in hierarchical environments but that in unstable environments a conservative approach in contrast to change-oriented management is more effective (Frederickson et al., 2016). More generally, they criticise that the fundamental assumptions of NPM, in particular contracting out for efficiency and change-oriented management style, lack empirical support (Frederickson et al., 2016). Despite the critique, NPM is a widely accepted and applied form of administrative theory and constitutes an appropriate tool for describing service delivery in stable environments. However, the explanatory power of NPM is limited for fragile environments.

### 2.2 State Fragility

State fragility refers to a continuum of situations in which a state lacks state authority and struggles to provide some (weak and failing states) or all (failed states) of the aforementioned core state functions (Hirschmann, 2016). The situation of state fragility can therefore be understood as a framework for different forms of state failure, state collapse and state reconstruction (Hirschmann, 2016). State fragility is a process that can be recognized before a complete state failure by "the rise of local or religious authorities, the [...] building of parallel structures, and the emergence of additional power actors/players" (Hirschmann, 2016, pp. 29–30).

Around the world, fragile states either result from the breakdown of artificial (former) socialist states, forced integration, permanent undesirable developments by the system of government or the construction of artificial states through former colonial rule (Hirschmann, 2016). The latter group pertains mostly to states in Africa and the Middle East, where demarcations through e.g. the Sykes-Picot Agreement between the former colonial rulers France and England constituted the country borders known today.

Even though there is not one precise definition of state fragility, the OECD offers a widely accepted one. A *fragile state* is characterized by a lack of capacity (effectiveness) and/or willingness (legitimacy) to perform key government functions such as meeting the population's basic needs, maintaining security and enabling economic development (François & Sud, 2006; OECD, 2008). Capacity thereby refers to the ability to mobilize resources and to execute territorial control. It is also related to the effective exercise of political power, economic management and the administrative capacity for policy implementation. Willingness refers to the "political commitment to policies supporting human welfare" and is reflected in an inclusive implementation approach, while legitimacy "concerns the sources of support for the state" (OECD, 2008, p. 14). These structural weaknesses do not go unnoticed but have a severe effect on the service delivery of fragile countries. The assessment of how state fragility affects service delivery outcomes is therefore at the core of this paper.

#### 2.2.1 Service Delivery in Fragile Situations

Services are delivered in a two-step process: Policy makers allocate them and service providers produce or implement them (OECD, 2008). Matland (1995) describes this process of policy implementation and distinguishes the top-down approach and the bottom-up approach. In the top-down approach, the policy designers are the central actors, creating policy at the central level. In the bottom-up approach, central actors devise government programmes at the macro-level of implementation and local organizers develop and implement the programmes at the micro-level of implementation, wherefore the service deliverers ultimately determine the policy. To differentiate this approach, Matland (1995) distinguishes between four policy paradigms. Administrative implementation in the context of low conflict and low ambiguity, political implementation in the context of high conflict and low ambiguity, symbolic implementation in the context of high conflict and high ambiguity, and experimental implementation in the context of low conflict and high ambiguity. Fragile settings are contexts of low conflict and high ambiguity, wherefore the policy outcome depends on the contextual conditions such as resources available and actors present (Matland, 1995). A compact between the policy maker and the implementer specifies the intended outcomes, while the client who uses the service does not partake in this compact. The users can hold the providers accountable either through the long route of accountability via the policy makers to which they express their (dis)satisfaction or directly to the producers with the short route (OECD, 2008). The long route of accountability requires an effective state and is mostly inoperative in fragile situations.

The OECD (2008) identifies three major problems that disrupt service delivery in fragile states. *First*, incentives for service delivery are negatively affected by a lack of government capacity and willingness, the breakdown of social order through conflict as well as through corruption or private capture of public resources. *Second*, service delivery tends to be fragmented because national policy

frameworks are missing. *Third*, poor infrastructure and technical and managerial capacity may hinder sustainable service provision as capacity development is subordinate to more urgent needs. The dilemma of trade-offs between short-term service delivery and long-term capacity development is also underlined by Brinkerhoff (2010).

The fragile context often leads to a situation in which capacity gaps in the public service delivery are filled by alternative modes of service provision to cater the humanitarian imperative and the needs of the population (Brinkerhoff, 2010; OECD, 2008). In fragile contexts with high will but low capacity, the international community may assign performance roles to non-state actors and bypass the state by directly contracting with alternative providers (Batley & Mcloughlin, 2010; Brinkerhoff, 2010). The non-state actors with which the central or local government contracts may be NGOs, International Organizations, Faith-based organizations, political movements or rebel groups. Moreover, various forms of client power such as community-driven development and co-production as well as market-based provision are possible (Baird, 2010; Brinkerhoff, 2010; OECD, 2008).

In these cases, service delivery depends on short-route accountability from service providers directly to users, which allows for greater participation and customization through voting with feet or with money (OECD, 2008). However, in fragile situations in which governments cannot guarantee political and economic stability or a legal system that would ensure contractual rights, formal contracting can hardly be effective (Batley & Mcloughlin, 2010). These alternative forms of service provision undermine the state's performance legitimacy that derives from service delivery. It decreases the population's trust in the state and disables the building of sustainable public-sector capacity (Brinkerhoff, 2010; Peace Security and Development Network, 2009). As the two tracks, service delivery and capacity development, require different strategies with different time-frames, this dilemma situation cannot be easily resolved.

Besides these impediments in service provision in fragile situations, service delivery is a broad term that implies a variety of delivery models depending on the characteristics of the environment and the service sector. For instance, in a context in which the desire for services is very heterogenous, decentralization improves service delivery by bringing decision making and beneficiaries closer together (Baird, 2010). Some sectors such as health contain an agreement on basic services and a shared practice of actions, which enable the import of international models of institutional reform and contracting-out the service delivery with NGOs. However, other sectors such as education present a different situation. The specifics of the education delivery in fragile state contexts are presented in the following section.

### 2.3 Education Delivery in Fragile Situations and Education in Emergencies

The education delivery in fragile contexts constitutes a special situation because "education is more than a basic service, in relation to fragility, education is at the same time cause, effect, problem and possible solution" (Kirk, 2007, p. 181). As previously mentioned, education promotes individual, family and community health and prosperity and empowerment. It can have a positive effect on the level of cohesion and group collaboration and can provide normalcy and opportunity for children (Kirk, 2007). At the same time, the omission and politicised commission of education can contribute to fragility.

The absence of education and later on job opportunities can contribute to instability as people with limited chances to improve their life turn to radical groups, who fill the vacuums in the public sector, or leave their home country (Kirk, 2007). The provision of education can be complicit in conflict and instability. Harmful practices and violence such as corporal punishment, bullying and sexual violence endemic within the school environment directly affect the development of children. More subtly, this environment denies children opportunities to develop their skills and to resolve conflicts peacefully. Schools can maintain the status quo by reproducing and exacerbating ethnic tensions and dominant forms of masculinity and femininity, thus contributing to instability and conflict (Kirk, 2007)

Since there is a reciprocal relationship between fragility and the omission and politicised provision of education, this leads to the imperative to prioritise sustainable education provision in fragile contexts that aims for the inclusion of diverse communities and the reconciliation of conflict. However, the education delivery in fragile contexts faces similar challenges than other sectors. The government lacks the capacity to provide education for the whole population and the education delivery is constrained by resource scarcity that leads to trade-offs with other sectors and between primary, secondary and tertiary education (Baird, 2010). This leads to a fragmented situation in which a variety of non-state actors are active in the education delivery.

A difficulty in the education provision by non-state actors is that unlike in the health sector, the education sector faces a "high degree of local content in the curriculum and different views on acceptable teaching practices" that make it "more difficult to contract out education services or provide them through non-state providers" (Baird, 2010, p. 13). However, if there is a tradition of non-state education delivery or there is mistrust in government agencies, non-state provision of education is a suitable option. When doing so, it is crucial to involve the community and to develop a compact by which "the government sets the curriculum and monitors the performance, in exchange for public funding" (Baird, 2010, p. 14).

This is recognized by a group of policy-makers, technical advisers and researchers who committed to this topic in the Inter-Agency Network for Education in Emergencies (INEE). The INEE

formulated the concept of *education in emergencies* to support decision-makers in enhancing the effectiveness and quality of their educational assistance in situations in which "children lack access to their national and community education systems due to occurrence of complex emergencies or natural disasters" (Kagawa, 2005, p. 494). Emergencies hereby specifically include human-made crises (e.g. war, internal conflict, genocide), wherefore the situation of war-induced Syrian refugees and the required restructuring of the Lebanese education system fall within this category. In these situations, other stakeholders such as multilateral organisations (e.g. UN), INGOs, and community-based organisations increasingly support the education sector with supplemental educational interventions and assume responsibility for the education provision (INEE, 2010). They execute education in emergency programmes which are promoted as a key element of the humanitarian response for both, host populations and refugees (Burde et al., 2017). They hereby operate on three dimensions of fragility sensitive education provision: (1) the equitable and effective delivery of education, (2) the content of education (e.g. the curriculum) and (3) the processes through which education takes place (e.g. teaching methods, management, administration systems and relationships) (Kirk, 2007). To achieve quality education that is available, accessible, acceptable and adaptable, the INEE formulated minimum standards for education in emergencies as a framework for good practice. These standards are divided into foundational standards, standards on access and learning environment, standards on teaching and learning, and standards on teachers and other educational personnel (INEE, 2010). By putting attention on community participation, the curriculum content, and the teacher selection process and training, these minimum standards go beyond the acute emergency reaction but provide guidance for capacity-sensitive education provision in fragile settings (Kirk, 2007).

#### 2.3.1 Defining Non-State Education Delivery

Due to various arrangements regarding the provision, financing and regulation, difficulties for the distinction between state and non-state spheres in education arise (Rose, 2007b). Privately owned schools may be subsidised by the government and the government may exert control over non-state education institutions through regulation. This impedes finding a clear definition of non-state education. UNESCO (2005b) in Rose (2007b) defines education as private if the institution is controlled and managed by a non-government organization, and distinguishes between government-dependent private schools and independent private schools, depending on whether they are more or less than 50 percent state funded.

Non-state education provision is not a new phenomenon. It started with missionaries and faithbased organizations that provided education in the Global South and later extended to NGOs focusing on vulnerable communities and for-profit private institutions focusing on domestic elites. In some countries, non-state education was established before a formal state-education system existed (Rose, 2007a). Nowadays, governments are mostly the main provider of education while non-state providers (NSPs) fill the gaps. In doing so, state and non-state actors can cooperate for education delivery in different ways, take responsibility for different aspects of service delivery and the interaction can range from informal give-and-take to formal contractual arrangements (Baird, 2010; Rose, 2007b).

In low-income countries and fragile settings, formal contracting of education service delivery by governments to non-state actors is rather rare as contracting requires an enabling policy and regulatory environment and a strong legal framework for education (Rose, 2007b). In these contexts, non-state provision is based on diverse cooperation forms that vary between fragile states that are in deterioration, post-conflict transition, arrested development or early recovery stages (Rose, 2007b). In a post-conflict transition such as in Lebanon, INGO provision and community-based approaches are dominant while engaging the government to build capacity. This cooperation between a state facing a fragile situation and a non-state education actor can take on five different forms. It can take place (1) within a government-led framework, (2) towards a government-led framework, (3) alongside a government-led framework, (4) outside of a government-led framework, and (5) within a global framework (Strategic Policy Impact and Research Unit [SPIRU], 2007). The third form works with multi-sectoral programmes that finance small-scale public investments with social funds. They require close working relationships with ministries and are usually implemented through community approaches in fragile settings (Berry, 2009).

While all NSPs seek to close a provision gap in the state school system, they differ in their focus groups. Community-based approaches first invest in the traditional links between the schools and the communities and later on transfer the school management and service provision to the communities (Rose, 2007b). This NSP is apparent when no other providers deliver education, requiring communities to set up their own schools. Faith-based organizations provide education to meet a more differentiated demand. They traditionally have a strong root in many countries through missionary schools and madrasahs<sup>i</sup> (Rose, 2007b). Philanthropic organizations, charitable organizations that are funded by individual donations or family trusts, mostly focus on poor communities. They require an economically well-off middle class or elite and are therefore less present in low-income countries (Rose, 2007b). Commercially-oriented private entrepreneurs often provide education services for the middle and upper class. Hereby, the distinction between for-profit and not-for-profit motivations is blurry. There are contract schools which are publicly funded and owned but managed by the private sector and within the public education system, some service aspects may be contracted out to private companies (e.g. quality assurance services) or the private sector may contribute with education inputs (e.g. textbooks) (Rose, 2007b). However, these marked-based provisions of education often lack ownership and are likely to be underprovided since the provision is not economically profitable for the private sector and the civil society (OECD, 2008).

Also, NGOs operate as non-state education provider and hereby focus on hard-to-reach groups which require alternative service delivery models and small-scale innovative approaches (Rose, 2007a). As refugees are a hard-to-reach group, NGO provision of education is the focus of this paper. These non-formal approaches to education via (more or less formal) contracting with non-state providers require a redefinition of education not as "time-bound, place-bound process confined to schools and measured by years of exposure" but rather "equating education with learning, regardless of where, how or when learning occurs" (Rose, 2007a: 6). Non-formal education can thus be seen as "organised, systematic, educational activity carried out outside the framework of the formal system" (Rose, 2007a: 6).

#### 2.3.2 Education Delivery Through NGO-Provision

Since the 1980s, NGOs and INGOs are increasingly involved in the education sector in fragile environments either through advocacy, supporting the quality of government provision or via providing education services themselves. Their education programmes can be distinguished into two groups. The first group of initiative integrates the education service with the state system, while the second group runs the educations service as alternative to the state system (Rose, 2007a).

The first type aims at integrating the non-state provision into the state system in the long run. These NGOs operate similarly to the state system by using e.g. the same curricula. The involvement of the community is usually limited, and the education service is seen as temporary, gap-filling solution as a response to a lack of government resources (Rose, 2007a). The NGOs mostly work on the community level by mobilising and sensitizing parents and local actors to improve the education conditions and encourage demand for schooling (Rose, 2007b). Also, innovative practices are piloted for government provision. The second form of NGO-provision of education is operating as an alternative to the state system. Alternative NGO provision seeks to meet the needs of excluded and vulnerable communities such as street children and refugees who need more flexible education delivery models<sup>ii</sup>. However, in fragile contexts in which donors cannot work through governments, NGO provision may sometimes not be an alternative but the only form of education provision (Rose, 2007b). Therefore, NGO education programmes are increasingly seen as complementary to state provision. Thus, small-scale local level involvement is often scaled-up and integrated with the government education provision, which requires a stronger alignment between NGO and government provision to allow a transition from one system to the other.

Most NGO involvement in the Global South draws on international donor funds. The cooperation between donors and NGOs can take place in three different modes. (1) Donors can finance projects through the government but make the involvement of NGOs in the delivery of the project compulsory (Rose, 2007b). (2) Donors can provide funds directly to a local NGO. This is the case when

there is low political will and/or capacity of states to manage contracts but there is a civil society. (3) Donors can provide funds directly to an INGO which in turn works with local NGOs (Rose, 2007b). This is done when there is low political will and low civil society capacity. This is mostly the case in conflict-affected states and fragile situations in which the donors do not intent to legitimize an unwilling regime, there is concern that the aid is not effectively managed by the public institutions, or there is an urgency to act in emergencies (Rose, 2007b). Bypassing government bureaucracies raises question for the sustainability of the involvement and possibly creates tensions with the state, also because they stand in competition with the government for donor resources. On the other hand, NGO provision is sometimes more accepted by the government as it constitutes a middle way between state and market provision and thus does not threaten the government monopoly on formal education (Rose, 2007b).

#### 2.3.3 Conceptualization of the Effectiveness of Education Delivery

With the emergence of the fragmented service delivery in the education sector, mechanisms that keep educational interventions in fragile states effective come to fore. Effectiveness is a contested concept that is mostly associated with performance outcomes. According to Rainey (2014) the basic performance assumption is that a public organization performs better if it defines clear goals and measures progress against them. This means firstly that effectiveness is closely connected to goal achievement and secondly that this approach requires monitoring instruments that allow for the measurement of the progress towards reaching that goal.

Any public or private organizations expresses its values and purpose (official goals) and seeks to meet specific ends with its procedures (operative goals). These goals are always *multiple*, as they are embedded in a set of goals, *conflicting*, because reaching a short-term goal possibly impedes a long-term goal, *organized in chains and hierarchies* in which one goal leads to another, turning them from ends into a means, and *shifting*, when the organization moves through different stages of its life cycle (Rainey, 2014). These goal complexities make it difficult to define what an organization's goal is and consequently to measure its achievement (Rainey, 2014). The focus on goals implies a rational orderly management process while "managerial decisions may be more turbulent, intuitive, paradoxical and emergent than a rational goal-based approach implies" (Rainey, 2014: 156). Despite the search for alternatives, assessments of effectiveness revolve around the goal approach in different variants.

One variant of the goal approach is looking at effectiveness on a systemic level by assessing the goals achievement of the whole network rather than of a single organization within the network. Networks hereby refer to "structures of interdependence involving multiple organizations or parts thereof, where one unit is not merely the formal subordinate of the others in some larger hierarchical arrangement" (Rainey, 2014: 169). For negotiated networks such as the fragmented education provision in fragile states, this would require measuring effectiveness from the government or international donor

that funds the service to the network of private and non-profit organizations that provide them (Rainey, 2014). Another prominent variant of the goal approach is to measure effectiveness based on criteria or goals that presumable all organizations must pursue (e.g. productivity, flexibility, adaptability) instead of determining specific goals for a specific organization. Further research on the goal approach in the education sector was carried out by Cameron (1978) who measured effectiveness mainly by drawing on perceptual criteria and measures reflecting the interest of students (Rainey, 2014). He found that even within the same sector there are conflicting criteria that weight against each other and that the institutions follow "different patterns of effectiveness" (Rainey, 2014:165).

All of the goal approach variants require monitoring instruments to assess to what extent the goals are achieved. These instruments need to follow a multidimensional approach covering system design, inputs, content process and outcomes. The instruments should be precise and measurable (United Nations Educational, Scientific and Cultural Organization [UNESCO], 2017). The assessment can be undertaken by an external actor or internally as long as the independent judgement can be guaranteed. For the education sector, monitoring mechanisms can stretch from assessments of student and teacher performance, condition of the school infrastructure, financial performance, organizational and administrative performance, to large-scale assessments on the country level such as the PISA study (UNESCO, 2017). Monitoring is integral for knowledge management and for identifying areas of opportunity and areas of intervention to improve performance outcomes.

#### 2.3.4 Conceptualization of the Barriers to Education

In the fragmented education delivery in fragile settings, NGOs seek to fill the capacity gaps of state school-systems to reach excluded and vulnerable communities. To achieve this goal and effectively provide education, the factors why students struggle to access formal education need to be outlined. These factors can be conceptualized as barriers to education that need to be overcome in order to provide education for marginalized communities such as refugees.

The barriers to education can be categorised into supply-side and demand-side barriers which interact with one another (Justino, 2016). Among the demand-side barriers are the direct and indirect costs of education and coping strategies of child labour and child marriage, psychological stress and trauma and being over age for the grade level.

**Cost of Education:** The most frequently mentioned barrier to education is the prohibitive cost of education (Wofford & Tibi, 2018). The exposure to violence or displacement leads to severe economic loss and forces refugees to live in extreme poverty. Parents may have to prioritize economic, social and psychological struggles above the education of their children (Wofford & Tibi, 2018). These circumstances impact on the household ability to send children to school as direct education costs such

as school fees and indirect costs for uniforms, books, certificates, lunch and transportation cannot be borne (United Nations Development Programme [UNDP], 2009).

**Negative Coping Mechanisms: Child Labour and Child Marriage:** In response to the economic vulnerability, children are required to assist in income generation activities to support their families. Child labour is a widespread coping strategy for households who face adverse economic shocks. The need to work impedes the education opportunities for children and often leads to school drop-out (Justino, 2016). These financial constraints are interlinked with the supply-side barrier of insecurity. In areas of high insecurity, parents withdraw their children from school in fear of (sexual) violence or rape (Justino, 2016). Parents are especially unlikely to allow their daughters to attend school if they fear violence on their way to or inside schools (Kirk, 2007). Securing safety through e.g. transportation poses an additional financial burden, wherefore often other coping mechanisms are deployed. The most frequent strategy to protect girls and to release financial burden is early marriage, which is affiliated with premature school drop-out (Kirk, 2007).

**Psychological Stress and Trauma:** Another demand-side barrier is psychological stress and trauma (Justino, 2016). Conflict and displacement impact on the cognitive abilities of children and lead to mental health issues that hamper school attendance and school performance (Aydin & Kaya, 2017). Resettled children may be dispersed across schools in a district or region depending on available placements. This leads to psychosocial isolation and mental health issues that add to the trauma of displacement and war (Wofford & Tibi, 2018). While socially integrated students are less likely to drop out of school, low-quality interpersonal relationships are attributed to higher likelihood of school drop-out (Bademci, Karadayı, & Vural, 2016).

**Over age for Grade Level:** Another factor that hinders children from accessing education is being over age for the grade level. Displacement causes a disruption of education for a significant amount of time. When refugee children want to continue their education, they are often too old for their grade level. Older students who have missed years of schooling are embarrassed to be schooled with significantly younger children and skip school or drop out (Sunny et al., 2017; Visconti & Gal, 2018b).

Next to these demand-side barriers, there are also supply-side barriers that need to be overcome to create an enabling environment. Among the supply-side barriers are a lack of language skills/curriculum, a lack of documentation and discrimination and violence.

Lack of Language Skills/Curriculum: Displaced children face difficulties in attending school because their native languages often differ from the language of instruction in schools. Therefore, a lack of language skills prevents them from participating in schools, which leads to repetition of grades or

school drop-out (Aydin & Kaya, 2017; Justino, 2016; Karam, Monaghan, & Yoder, 2017; UNDP, 2009; Visconti & Gal, 2018b; Wofford & Tibi, 2018).

Lack of Documentation: Another supply-side barrier to education resulting from displacement is a lack of documentation. Children mostly need to show proof about the completion of the previous grade in order to enrol in schools (Justino, 2016). Due to displacement, this proof cannot be supplied, wherefore they either cannot enrol at all or not in their grade level as they are denied taking exams for higher levels (Visconti & Gal, 2018a). This category refers to legal documents such as residency permits that deter enrolment because children fear reporting of their irregular situation in or on their way to school (UNDP, 2009).

**Discrimination and Violence:** Another important factor that prevents refugee children from accessing education is discrimination and violence. When migrants or refugees enter a country's education system, scarce resources are stretched, leading to resentment between the host population and the refugees (Wofford & Tibi, 2018). As a result, the integration into foreign classrooms is difficult and bullying among refugees and between refugees and native students frequent (Visconti & Gal, 2018a). Bullying is a proactive type of aggression that aims at hurting others though physical assault, verbal harassment and via indirect means such as anti-social behaviour (Bademci et al., 2016). This sort of discrimination also encompasses xenophobic attacks, neglect or exclusion by teachers and school administers (Karam et al., 2017; UNDP, 2009). Coercion and (corporal) punishment to deal with inappropriate student behaviour instil fear within the children and lead to their absence or drop out (Bademci et al., 2016).

Since many of these barriers interact with one another, refugees constitute a multiply marginalized group (Rose, 2007b). They reside in poorer areas with higher insecurity and where lesser schools are available. They are affected by linguistic disadvantages and often face discrimination based on ethnicity and gender. Barriers to education are thus best viewed from an intersectional perspective (Ontario Human Rights Commission, 2001).

This chapter provided the theoretical framework for the thesis. Section 2.1 elaborated on New Public Management as a framework theory for service delivery in fragile environments, after which section 2.1.1 detailed the characteristics of contracting out for efficiency. Section 2.2 described the characteristics of fragile settings and section 2.2.1 explained how they disrupt service delivery. Section 2.3 zoomed in on education delivery in fragile situations. It outlined the connection between fragility and education and presented emergency education as concept deployed in crisis situations. Section 2.3.1 highlighted the fragmented education delivery in which non-state actors fill the capacity gaps of state education delivery. Then the education provision by NGOs was presented in more detail and their specific target group of hard to reach communities such as refugees was identified in section 2.3.2.

Subsequently, section 2.3.3 conceptualized the effectiveness with a monitored goal approach. Lastly, section 2.3.4 presented the conceptualization of the supply- and demand-side barriers to education for refugees that need to be overcome to enable their access to education in fragile contexts.

## 3. Data and Documents

This chapter presents the deployed methodology consisting of research design, the data analysis method and the limitations that occur in the research process. It then continues with a clear operationalization of the concepts that are described in the theory section.

### 3.1 Methodology (Research Design, Data Analysis Method and Limitations)

The thesis examines *how and to what extent selected education projects undertaken by non-state service providers address the factors inhibiting the access to education for Syrian refugees in Lebanon effectively.* Taking the country and displacement context into account, the study seeks to evaluate the effectiveness of the non-state education provision in order to give recommendations on how education can be better targeted at marginalized and displaced groups. The thesis is a case study that focuses on one single instance, namely on the social phenomenon of lacking access to education for Syrian refugees in Lebanon (Babbie, 2013). Unlike the clear typology of research designs in quantitative research, in the context of qualitative research the research design "should be a reflexive process operating through every stage of the project" (Maxwell, 2008, p. 214). A case study design refers to describing the original view of the problem, the methodological and theoretical commitments and how these affect the research and vice versa (Maxwell, 2008).

With a longitudinal case study approach, the study presents the evolution of the emergency education response to the displaced Syrians since the crisis outbreak in 2011, thus looking at the case of non-state education provision in a fragile setting over time. The research design thereby follows a process tracing approach which analyses the qualitative interferences, namely the NGO and INGO education programmes that are sought to tackle barriers to education (Blatter & Haverland, 2012). By examining the qualitative interferences as units of analysis, the process from causal factor to outcome can be traced back. The process tracing thereby helps to identify steps for intervention by examining a combination of divergent factors in the process that lead to a specific outcome (Blatter & Haverland, 2012). The factors refer to the barriers to education which are observed via a secondary data analysis of project reports, which are the units of observation. Deploying the process tracing approach enhances the internal validity by enabling an assessment if for instance one change in approaching refugee children led to more/less effectiveness of one project than another. In order to provide temporal order, density and depth of the description of the causal process, storylines with the major steps and detailed descriptions of important moments are important (Blatter & Haverland, 2012). The appendix provides

a portray of Lebanon's historical background to develop a coherent storyline. The emergency education response for refugees constitutes an influential case within the education sector as it "fits the overall model [...] [and] might affect the overall set of findings for the whole population" (Seawright & Gerring, 2008, p. 303). Since the study is conducted with one case, the education sector in Lebanon, it is unlikely to play an influential role in terms of generalization, but the case of Lebanon provides in-depth understanding of the refugees' precarious situation, which is equally important (Seawright & Gerring, 2008).

Maxwell (2008) refers for qualitative research to transferability instead of generalizability. He developed an *interactive model of research design* with the components goals, conceptual framework, research questions, methods and validity. The components are subject to reciprocal effects and address different issues relevant for the coherence of the study (Maxwell, 2008). In this paper, the intellectual goal is to describe how and to what extent selected education projects undertaken by non-state service providers address the factors inhibiting the access to education for Syrian refugees in Lebanon effectively. This approach preserves the individuality of the situation and leads to a better understanding how actions and events are shaped by the unique circumstances in which they occur (Blatter & Haverland, 2012; Maxwell, 2008). The conceptual framework of this study refers to the key concepts outlined in the theory section. The *research question* re-connects to the other components of the design, such as the concepts and the goals, focuses the study and provides guidance on how to conduct it (Maxwell, 2008, p. 229). The *methods* that are used in this paper can be characterised as *intensive* research design as it focuses on the "particular phenomena studied" and the "internal validity and contextual understanding [...] [of] the processes that led to specific outcomes" (Maxwell, 2008, p. 233). The paper is interested in the particular phenomenon of non-state education provision for Syrian refugees in Lebanon and focuses on the differences in the conducting of the respective projects that lead to effective or ineffective outcomes.

In qualitative research, data analysis consists mostly of categorizing strategies through coding which enables to "rearrange and compare between things in the same category and between categories" (Maxwell, 2008). In this study, the arguments are derived from academic and grey literature via desk research. Precisely, data is extracted from project evaluation reports on educational projects, which are structured and coded for comparison. Due to time constraints, there is no opportunity for conducting expert interviews or surveys with the affected refugee population. Drawing on the conceptualization of effectiveness and barriers to education from section 2.3.3 and 2.3.4, the projects are analysed regarding the extent to which they address the barriers to education effectively. The conclusions of this paper will be drawn on the basis of the whole examination of the influential case. Possible validity threats resulting from researcher bias are sought to be countered by collecting information from a diverse pool of authors

and reliable organizations (triangulation) and by searching for discrepant evidence and negative cases (Maxwell, 2008).

#### 3.2 Operationalization and Data Collection Methods

The main concepts in this paper are *service delivery* (access to education), *state fragility*, *education in emergencies*, *effectiveness*, and *barriers to education*.

*Service delivery* can generally be assessed at the sector level. This study focuses on education delivery in Lebanon which is operationalized with the level of education enrolment that can be derived from the Ministry of Education and Higher Education (MEHE) for formal education. Enrolment data on refugees can be derived from the Vulnerability Report by UNICEF, UNHCR, and World Food Programme (2015, 2016, 2017; 2017). Also the completion rates, literacy rates and numeracy rates provide insights in the level of education delivery (The Fund for Peace, 2018). These rates can also be extracted from MEHE or from independent sources of external parties (United States Agency for International Development [USAID], 2018). *State fragility* assessed via the Fragile States Index by the Fund for Peace which can be used to receive numerical data (The Fund for Peace, 2018).

*Education in emergencies* can be approached via its main task, the supplemental educational interventions. These are operationalized with non-state education programmes, e.g. community-based homework groups and basic numeracy and literacy programmes (Deane, 2016). As formal education provision is largely unavailable for refugee children, non-formal education (NFE) initiatives are considered effective to the extent that they adapt education to the realities of the displaced children (Deane, 2016). The necessary information is derived from project reports of NGO or INGO education project reports. As the data availability of NFE delivery is limited, other research studies circumvent this problem by creating measurement methods such as monitoring school visits (Echessa & Pinnock, 2012). Since this is not possible for this thesis, the availability of project reports is a prerequisite for the data collection.

Besides the general screening of possible non-state actors through search engines, the researcher consulted the Operational Portal Refugee Situations by UNHCR that lists the partners of UNHCR per country and sector. Unlike for the other countries in the region, the website does not list the partners in Lebanon. The researcher looked for non-state education actors on the websites of the Lebanese government, UNHCR, UNICEF and UNDP as well as on lists provided by the American University of Beirut and the Centre for Educational Research and Development of Lebanon. Over the course of the thesis, the researcher identified possible non-state education projects and looked for the public availability of project reports (exclusion criterion). The researcher contacted promising education projects<sup>iii</sup> that provided descriptions of their approach but no publicly available evaluation reports. After

non-response, follow-up e-mails were sent. Until the end of the research process, no response was delivered by the contacted NFE providers apart from Basmeh & Zeitooneh which responded that no evaluation report can be sent. Dealing with the limited data availability, the researcher decided to focus on INGO projects. Via the Lebanon Humanitarian INGO Forum (LHIF) three INGO projects could lastly be selected. In order to find a fourth project, the academic and grey literature was screened (again) and all promising references to other project reports were followed up on. The researcher faced the obstacle that the other researchers had access to internal evaluation reports. In order to uphold a minimum of four education projects for the analysis, an outreach campaign for education enrolment was selected despite not consisting of NFE itself. To compensate for the small selection of projects, the researcher included the information found during the search for projects in the overview on NGO and INGO education provision in section 4.2.1. Yet, this overview is not representative.

When analysing these project reports, the projects are considered *effective* (1) if they deploy monitoring mechanisms and (2) if they achieve the goals they have set in the beginning. The monitoring operationalized via assessments mechanisms are at the students, teacher and administrative/organizational level (UNESCO, 2017). The goal achievement is operationalized as the project output, as indicated in the project reports. Monitoring and evaluation data is extracted from project evaluation reports and from ActivityInfo, the response planning and monitoring platform UNICEF uses.<sup>iv</sup> Rainey (2014) argues that relying on self-described effectiveness in project reports is only the second-best option and that personal data collection in the field is preferable.

The demand-side barriers to education that consist of the cost of education, negative coping mechanisms, psychological stress and trauma, and being over age, as well as the supply-side barriers to education consisting of the lack of language skills/curriculum, lack of documentation, and discrimination and violence are operationalized in the following way. The cost to education is assessed via the presence of school feeding programmes, the provision of stationary and books, the provision of transport or school uniforms, and the abolition or waiving of tuition fees, which increases the net attendance ratio and the primary school completion rate (Jennings, 2016; Langsten, 2017). The negative coping mechanisms can be operationalised with conditional or unconditional cash transfer programmes that transfer cash to families when they fulfil certain activities such as school enrolment (Justino, 2016; Masterson, 2016; Morais de Sa e Silva, 2015). Interventions that reduce the poverty of families is also operationalized as targeting negative coping mechanisms as poverty contributes to absence in school due to the need to work. Another operationalization is the restructuring of schools so that work can be reconciled with education (Save the Children, 2009). Regarding the coping mechanism of early marriage, the involvement of family and community and higher shares of female teachers are sought to contribute to prioritizing education (Save the Children, 2009). The barrier of psychological stress and trauma is operationalized with the access to psychological counselling or psycho-social activities (Aydin & Kaya, 2017; McLaughlin & AL-Hroub, 2016; Shuayb, Makkouk, & Tuttunji, 2014). The barrier of being *over age* for the grade level is operationalized through repetition rates, indicating that age-forgrade heterogeneity contributes to grade repetition (Sunny et al., 2017). The barrier of *lack of language skills/curriculum* is operationalized as presence of additional language support, the development of multilingual teaching material, the presence of literacy and numeracy tutoring, homework support, or remedial support (McLaughlin & AL-Hroub, 2016; Shuayb et al., 2014; Wofford & Tibi, 2018). The *lack of documentation* is operationalized with easing enrolment processes and registration for exams. *Discrimination and violence* are operationalized as non-violent teaching practices, the presence of an inclusive classroom approach, and the absence of bullying, (corporal) punishment and politicised curricula (Awada, Diab, & Faour, 2018; Echessa & Pinnock, 2012; Save the Children, 2009).

## 4. Analysis

The analysis chapter prepares the answer to the central research question of *how and to what extent selected education projects undertaken by non-state service providers address the factors inhibiting the access to education for Syrian refugees in Lebanon effectively* by analysing and answering three sub-research questions. For each sub-question, first the data is reported and second the data is discussed in the light of the theoretical framework.

## 4.1 What are the Factors that Constrain the Access to Education for Syrian Refugees in Lebanon?

The first sub-research question asks "*What are the factors that constrain the access to education for Syrian refugees in Lebanon?*". In order to answer this question, section 4.1.1 outlines the Syrian refugee crisis and the response in Lebanon, section 4.1.2 sketches Lebanon's resulting socio-economic situation, section 4.1.3 presents the Lebanese education sector, and section 4.1.4 describes the adjustments made in the education sector to accommodate the refugees. Section 4.1.5 answers the first sub-research questions by identifying the factors that constrain the access to education for Syrian refugees and viewing them in the light of theory. This chapter is best understood embedded in Lebanon's historical background, which is provided in the annex.

#### 4.1.1 The Refugee Crisis and the Response in Lebanon

The Syrian war is the most intricate conflict in the world right now, resulting in an estimated 400.000 civilians being killed and the majority of the country's infrastructure being destroyed (Dionigi, 2016). The semi-anarchical and warlord-ruled area leads to approximately 6.6 million internally displaced persons and around 4.7 million refugees who seek shelter in the region and worldwide, turning the Syrian conflict into an unprecedented incident of forced migration in the modern history of the Middle East (Dionigi, 2016). Given the geographic proximity, the shared history and language as well as family connections and trade relations, Lebanon is one of the main destinations of Syrian refugees.

With more than 1.5 million Syrian refugees, Lebanon hosts the highest per capita share of refugees worldwide, with refugees exceeding 30 percent of the country population (MEHE, 2014; UN Lebanon, 2017). The arriving refugees consist of Syrian nationals, Lebanese returnees and Palestinian refugees who settled in Syria. Because of the previous conflicts, international organizations already established a presence in Lebanon for dealing with Palestinian refugees. Therefore, international organizations including the United Nations (UN) agencies took the initial lead in responding to the Syrian refugee crisis in the absence of a national coordination strategy (Boustani, Carpi, Gebara, & Mourad, 2017). Lebanon opened its borders in a laissez-faire approach, but neither formulated a mandate for the many humanitarian organizations present nor developed a national response strategy to the crisis, because Lebanon does not view itself as a country of asylum and is not a party to the 1951 UN Refugee Convention (Janmyr, 2016).

As the repercussions of the Syrian refugee crisis not only affected Lebanon but the whole region, UNHCR and UNDP developed several regional response plans for the humanitarian emergency response to the Syria crisis. In 2012, the plans resulted in the Regional Refugee and Resilience Plan (3RP) that introduced a long-term component for the countries most affected by the crisis (Lebanon, Jordan, Iraq, Turkey and Egypt) (United Nations High Commissioner for Refugees [UNHCR] & UNDP, 2017, 2018). Lebanon's country version of this plan is the Lebanon Crisis Response Plan (LCRP) which was drafted in the end of 2014 and updated at the end of 2015 for 2016 (GoL & UN, 2015, 2014, 2018). The LCRP is designed to ensure humanitarian assistance and protection, to strengthen the capacity of the national and local service delivery systems and to reinforce Lebanon's economic, social, environmental and institutional stability (GoL & UN, 2015). The initial plan was later transformed from an emergency response into a longer term strategic framework for 2017 till 2020 (GoL & UN, 2018). Over time, the Lebanese government, especially the Ministry of Social Affairs, increasingly participated in the development of these plans. The LCRP is divided among sectors and regions and each task is led by UNHCR, a sector-specific agency (e.g. UNESCO for education), and a national ministry (Ministry of Education and Higher Education (MEHE) for education) (Boustani, Carpi, Gebara, & Mourad, 2016). The sector initiative for education is the Reaching All Children with Education (RACE) plan, which aims at providing access to quality formal and non-formal learning opportunities for children in a safe and protective environment (MEHE, 2014).

Regardless of the efforts on multiple levels, the influx of Syrian refugees influences the situation in Lebanon. Since the beginning of the refugee crisis in Lebanon, political, social and economic challenges that mostly existed in the country prior to the war in Syria have been exacerbated, leading to worsening living conditions for both refugees and host communities. Therefore, the following section briefly describes the impact of the Syrian refugees on Lebanon in economic, social and political terms.

#### 4.1.2 The Lebanese Socio-Economic Situation

Regarding the economic side, the main consequences of the Syrian refugee crisis are an increased expenditure due to the population increase in combination with economic losses in the main sectors (UN Lebanon, 2017). Despite the fact that Lebanon is still an upper middle income country, Lebanon's economy faces negative repercussions of the Syria conflict (USAID, 2018). The gross domestic product fell from 8 percent in 2007-2010 to 1.5 percent in 2013 (MEHE, 2014). The Syrian conflict disrupts key growth drivers such as banking, construction, tourism and the service sector, and affects the import and export of commodities over the land route, leading to increased prices. At the same time, the Syrian refugees impact the labour market, leading to decreased wages due to more competition (Cherri, Arcos González, & Castro Delgado, 2016). Living costs and the inflation are rising. This development pushed approximately 170.000 Lebanese families into poverty (MEHE, 2014). While prior to 2011, 28 percent of the Lebanese population were living under the poverty line of US\$3.84/day and 8 percent under the extreme poverty line of US\$2.4/day, the overall Lebanese poverty has increased to 32 percent (GoL & UN, 2015). The economic inequalities are deepened and negatively impact on social relationships, ultimately posing a threat to the social cohesion and the country's stability.

With regard to the social dimension, the settlement of Syrian refugees plays an important role. Given Lebanon's urban character and the strict no-camp policy following the conflicts in Palestinian refugee camps, the majority of the displaced Syrians seek shelter in Lebanon's most vulnerable and poorest communities along the poverty belts of large cities due to lower living costs (Oxfam, 2015). 86 percent of the Syrian refugees live in 242 communities where 66 percent of Lebanese live on less than US\$4 a day (UNICEF et al., 2017). These are the traditionally deprived regions in the North (30 percent) and in the Bekaa valley (34 percent) (MEHE, 2014). There are over 400 informal settlements hosting 12.5 percent of the registered refugees (MEHE, 2014). The influx of refugees in the communities exacerbates the difficult socioeconomic situation and challenges the coping capacity of the Lebanese host community. The distribution of aid to Syrians fuels tensions with the poverty-stricken Lebanese communities (Turner, 2015). In addition, the mostly Sunni Muslim Syrian refugees upset the sectarian balance in Lebanon.

On the political side, the influx of Syrian refugees meets a divided stance between the groups who support the opposition in Syria and those who support Assad, which makes refugees subject to political contention. This antagonism is also present in the government, leading to political deadlocks in decision-making on refugee issues, as the allocation of public expenditure follows confessional lines (Janmyr, 2016; Salti & Chaaban, 2010). The Syrian refugees add to an already tense situation in the public service delivery. For instance, municipal spending on waste disposal has increased by 40 percent and 92 percent of untreated sewage contaminates water sources, because the increased service demand exceeds the capacity to meet it (Cherri et al., 2016). The infrastructure and public health care systems

struggle because they lack staff, equipment and medication (Cherri et al., 2016). In addition, the nocamp policy as well as the absence of an official census impede the appropriate targeting of refugees with the needed services as it is unclear how many people are in the country and were they reside. This creates a challenge for NGOs and UN agencies in terms of outreach, accessibility, and registration with the municipalities (Boustani et al., 2017).

The aforementioned situation sheds light on the structural and systemic problems that existed in Lebanon prior to the refugees and shows how the Syrian refugee crisis aggravates them. The socioeconomic and political impact of the Syrian refugee crisis is enormous on a country that still recovers from years of conflict and struggles to mitigate sectarian tensions. Especially the public service provision is struggling to keep up with the sudden and large increase in demand, causing a decline in public service access and quality, and a rise of tensions with the marginalized Lebanese poor. One of the most problematic sectors is education, as it used to not be covered by the humanitarian response even though it is crucial for a generation of children and subsequently the future of Syria and the region. Therefore, the following section pays special attention to the education sector.

#### 4.1.3 The Lebanese Education Sector

Similar to the overall overburdening of the Lebanese public sector, the Lebanese education sector was already in need of reform prior to the Syria crisis (Cherri et al., 2016). The Lebanese school system is divided into elementary and intermediate education, catering for the age group 6-14, and secondary education, catering for 15-17 year olds (UNESCO et al., 2014). The Lebanese school system consists of public, subsidized and private schools.

The main problems in the Lebanese education sector are the small proportion of public provision and the low quality of the provision. The public system caters only for 30 percent of the Lebanese student population (Human Rights Watch, 2016). These are mostly the communities with the lowest socioeconomic characteristics that are attracted through the tuition free education but can afford the related education costs in registration, supplies, uniforms and transportation. As only 44 percent of the employed teachers are in the public sector, the student-teacher ratio is high (MEHE, 2014). Due to the overcrowding of the classrooms and the low quality of teaching, public schools show a 10 percent lower academic outcome in test scores in comparison to private schools and their repetition rates, a proxyindicator for learning, lay at 19 percent (MEHE, 2014). Moreover, the school infrastructure requires rehabilitation in many cases, especially for water and sanitation facilities. In 2011 only 78 percent of public schools had access to piped water systems (MEHE, 2014). One third of the school buildings are not state-owned and thus require MEHE to rent more than 400 schools, which poses an additional financial burden on the budget for public education (MEHE, 2014). The Lebanese government spends 2.48 percent of the GDP on education, thereby remaining far below the regional average of 4.2 percent (USAID, 2018). However, it put significant effort into increasing the number of public schools and improving their quality. But despite the efforts the progress remained inadequate, wherefore many students drop out of school and parents who can afford it chose to send their children to private schools due to perceived better performance.

Because of these constraints in the public school-system, non-state education provision forms an important pillar in the Lebanese education sector. Besides the public schools, there are free private confessional schools provided by faith-based organizations, which are sometimes subsidised by the state and usually teach the Lebanese curriculum (MEHE, 2014). Additionally, there are non-free private schools provided by for-profit organizations in exchange for tuition fees, teaching a specific curriculum. The private schools account for more than half of the enrolled Lebanese students (UNESCO et al., 2014). Furthermore, there are UNWRA schools for Palestinian children at primary level that are free and teach the Lebanese curriculum (MEHE, 2014). This precarious education situation has been exacerbated by the Syrian refugee crisis.

#### 4.1.4 The Adjustments in the Lebanese Education Sector

Of the population influx of 1.5 million officially UNHCR-registered Syrian refugees, an estimated 42 percent are school aged children between 3-18 years (MEHE, 2014). Of these 420.000 Syrian children an estimated 180.000 Syrian children and an additional 50.000 Lebanese children are in primary school age (6-14 years) (UN Lebanon, 2017). This leads to a tense situation with additional pressure on the underfunded education sector. It demands not only the increase of the education services but also more variety to cater specific needs and to compensate for negative impacts on the quality of the public education system.

In response to this emergency, the aforementioned Reaching All Children with Education I (RACE I) plan was adopted for 2014-2016 as the education sector part of the LCRP. A follow up plan, RACE II, was established for 2017-2021. RACE I builds on the "No Lost Generation" Strategy and a stabilization framework developed by the UN, the World Bank and the Lebanese government (GoL & UN, 2015). The RACE II outlines a multi-year comprehensive approach to education and covers the humanitarian response and long-term support (GoL & UN, 2018). The plan targets a population of around 460.000 Syrian children, 6.500 Palestine children from Syria, 6.500 Palestine children in Lebanon, as well as 81.000 vulnerable Lebanese children with an investment of US\$388.2 million (GoL & UN, 2015).

As part of the response to accommodate the additional students, MEHE introduced a variety of mechanisms for the public school-sector. Since 2013, it prompts 364 public schools to introduce a second shift approach with Lebanese children mostly attending in the morning and Syrian refugees attending from 2:00 to 6:00pm (GoL & UN, 2018). The afternoon shifts offer a modified syllabus which contains a condensed form of the Lebanese curriculum (Deane, 2016; GoL & UN, 2018). MEHE

facilitates the access to education via mandating public schools to enrol refugees regardless of their legal status and waiving school and book fees since 2015 (El-Ghali, Ghalayini, & Ismail, 2016; United Nations High Commissioner for Refugees Lebanon [UNHCR Lebanon], 2016). MEHE introduced a certified Accelerated Learning Programme as a pathway to formal public education for children from seven to 24 years who have been out of school for more than two years (MEHE, 2016). It teaches a condensed form of the Lebanese curriculum, mostly one grade of the national curriculum in three months (Deane, 2016; Norwegian Refugee Council [NRC], 2017). For children who do not qualify for the Accelerated Learning Programme, the Basic Literacy and Numeracy programme for ten to 18 year olds is offered (MEHE, 2016). However, on average only 35 percent of the children who attend the Accelerated Learning Programme transfer to the formal public schooling (GoL & UN, 2018). In line with these MEHE-accredited NFE programmes, MEHE also formulated a Non-Formal Education Framework that standardizes and centralised NFE programmes since 2016 (NRC, 2017). In addition, Syrian refugee children also receive subsidised formal education in many private schools and NFE programmes are offered in 205 informal settlements across Lebanon (MEHE, 2014).

However, these changes did not come without problems. The double-shift approach puts extra burdens on teachers and students and the government struggles with the costs of the Accelerated Learning Programme (Karam Foundation, 2016). Moreover, the influx of refugees in the most vulnerable communities adds to the overcrowded situation that existed prior to the refugee crisis. These conditions lead to rising tensions within the class and between students and teachers. The teachers are not prepared to manage the new needs and draw on violent coping mechanisms. In 2012, the incidence of violence in the classrooms was at 70.4 percent (MEHE, 2014). The overcrowding also contributes to the already high utilization and deterioration of public school infrastructure, which required rehabilitation already prior to the refugee crisis.

Until now there has not been a comprehensive impact assessment of RACE I that provides reliable data on measurable indicators such as enrolment and dropout rates. The data that is available shows considerable variation, depending on the actor publishing them and sometimes even in different documents provided by the same actor (Basmeh & Zeitooneh, 2017). Therefore, the numbers presented in the following section are sometimes optimistic estimates by the Lebanese government and sometimes conservative estimates by education actors such as NGOs, hoping to display a realistic middle way. Especially the data of the Vulnerability Reports of 2015, 2016 and 2017 are used because they are the result of a comprehensive assessment of the Syrian refugee situation in different areas with mixed methods (UNHCR et al., 2015, 2016, 2017).

Prior to the RACE I plan, the government efforts led to a refugee enrolment rate in public education in the school year 2012-2013 of approximately 40.000 students (13.5 percent), in the 2013-2014 school year 88.000 (30 percent) and in the 2014-2015 school year 106.000 (36 percent) (GoL &

UN). For secondary education, the enrolment rate laid at 2 percent in the 2012-2013 school year (MEHE, 2014). After the introduction of the RACE I plan, the enrolment rates for primary education (6-14 year olds) raised from 52 percent in 2015 and 2016 to 70 percent in 2017 (UNHCR et al., 2015, 2016, 2017). In the most deprived Bekaa region the enrolment rate raised from 36 percent in 2015 and 2016 to 59 percent in 2017 (UNHCR et al., 2015, 2016, 2017). In turn, this means that today a minimum of 30 percent of primary aged children are still out of school. Even though the enrolment rates are rising, the completion rates remain a challenge. In 2015, only 46 percent for primary education (UNHCR et al., 2015, 2017). The failure rate of Syrian children is twice as high as the one of their Lebanese peers. For secondary education (15-17 year olds) the enrolment rates stayed at around 5 percent from 2015 to 2017 (UNHCR et al., 2015, 2016, 2017). In 2015, 89 percent of the secondary school children were out of school, which decreased to 84 percent in 2016 (UNHCR et al., 2015, 2016). Also here the completion rate remains a challenge, with 11 percent for upper secondary education (grade 10-12) (UNHCR et al., 2017).

Even though the initiative reached significant amounts of children, the RACE I strategy has failed to reach its targets of providing around 530.000 students with education in the 2015/2016 school year (GoL & UN, 2015). This can mostly be drawn back to insufficient funding, weak coordination among government agencies and inadequate collaboration with NGOs (El-Ghali et al., 2016). MEHE estimated that the additional cost of teaching each Syrian refugee child is US\$600 per year and the budget of MEHE fell short on US\$140 million (MEHE, 2014). In addition to the high number of children who remain either out of school or drop out along the way, the children who remain in school are not always learning due to language barriers and poor quality of the curriculum and teaching. Given that a situation analysis of the Syrian youth in Lebanon by UNESCO et al. (2014) shows that 95 percent of them are willing to continue their education in Lebanon if possible, it is worthwhile to look at the factors that hinder Syrian refugee children from accessing or remaining in education.

## 4.1.5 Answer to the First Sub-Research Question: The Factors that Constrain the Access to Education for Syrian Refugees

In the education sector, soft security concerns of the refugee situation such as intra-communal tensions and marginalization can be addresses by responding to the particular needs of Syrian children (Deane, 2016). Therefore, it is important to know about the factors that still constrain Syrian refugee children from accessing education. In order to not only derive the barriers to education from one moment in time, the Vulnerability Reports of 2015, 2016 and 2017 are taken into account, thus providing persistent barriers to education. Additionally, further reports are used to increase the reliability of the factors identified. The barriers to education are divided into demand-side barriers and supply-side barriers as they require different forms of action to tackle them (Justino, 2016).

Among the demand-side barriers are the cost of education and accompanying negative coping mechanisms such as child labour and child marriage, psychological stress and trauma, and being over age.

**Cost of Education:** In numerous assessments the cost of education is identified as the main barrier to education across all age groups. Besides mostly waived tuition fees in public schools, costs arise for uniforms, supplies and transport, hindering refugees from accessing education (El-Ghali et al., 2016). 71 percent of the households whose children were out of school had an income of less than US\$300 (UNHCR et al., 2015). In line with that, 39 percent of children between 6-15 years and 36 percent of children between 15-17 years mention the cost of education as main barrier to education (UNHCR et al., 2017).

Negative Coping Mechanisms: Child Labour and Child Marriage: The precarious life circumstances of Syrian refugee children due to displacement make them one of the most marginalized groups who settle in the most vulnerable communities. It also leads them to living in poverty. Poverty forces people to adopt negative coping strategies such as child labour and other forms of exploitation. Families often rely on the children's income wherefore they deprioritize their children's education (Karam Foundation, 2016). Seasonal child labour is particularly prominent in the Bekaa, North and South, where the communities face a depletion of resources and a reduction of assistance (GoL & UN, 2015). 7 percent of the Syrian refugee children between 12-14 years mentioned the need to work as a barrier to their education (UNHCR et al., 2015). Among the 15-17 year old, this number rises to 13 percent (UNHCR et al., 2016). In some communities, the community leader (Shawish) also refuses children to go to school in order to send them to work (GoL & UN, 2015; United States Department of State, 2017). Besides child labour, poverty also leads to the negative coping strategy of early marriage. It disproportionally affects girls who are married off their families to release financial burden and supposedly provide for their safety in insecure environments by assigning them a husband. One in five girls aged 15 to 19 are married and 18 percent of them to a spouse that is ten or more years older (UNHCR et al., 2017). Among the 15-17 year olds, 7 percent report being married as a reason for being out of school (UNHCR et al., 2016).

**Psychological Stress and Trauma:** Displaced Syrian children that have witnessed or experienced violence, the destruction of homes or losses of family members are affected in their psychological well-being (GoL & UN, 2015). This is reflected in symptoms such as sleep disturbances, crying, screaming, bed-wetting, nightmares, clinginess and withdrawal (GoL & UN, 2015). The psychological well-being of children is aggravated by the dire living conditions and the tensions with the host community. Only 11 percent of Syrian refugee youth describe their state of mind as positive, while 30 percent feel anxious or depressed, 22 percent afraid, and 6 percent despondent (UNESCO et al., 2014). To 17 percent of Syrian refugee youths the thought of ending their life occurs often while 24

percent report to have that thought sometimes (UNESCO et al., 2014). However, no information was to be found on the extent the psychological stress and trauma cause them to drop out of school or prevent them from enrolling.

**Over age for Grade Level:** The displacement disrupted the education of many Syrian refugee children sometimes for years, making it difficult for them to resume their education when resettling. 24 percent of the children state their age as a reason for being out of school (UNHCR et al., 2015). Of the children enrolled in the first grade of primary school, 54 percent are two or more years older than the standard age of that grade (UNHCR et al., 2017). 11 percent of the primary students are between 3 to 5 years older than the standard age for that grade (UNHCR et al., 2017). For early childhood (pre-primary) education, the primary barrier for being out of school is that parents consider the children too young for education (UNHCR et al., 2017).

Next to the demand-side barriers, there are also considerable supply-side barriers who affect the access to education for Syrian refugee children. Among them are a lack of language skills and the difficulty of the curriculum, the lack of documentation, discrimination and violence in schools, and inadequate availability of schools in deprived areas.

Lack of Language Skills/Curriculum: The Lebanese curriculum differs greatly from the Syrian curriculum with regard to content and language of instruction (El-Ghali et al., 2016; Karam Foundation, 2016). In Lebanon, the languages of instruction are Arabic, French and English, with the latter two languages being taught as second languages and being utilized for the instruction in math and science (Karam et al., 2017). As French and English are only taught as foreign languages in Syria, this leads to insufficient literacy in these languages (Deane, 2016; GoL & UN, 2015). This makes it difficult to catch up even for children whose education was not interrupted. For children with an education gap, keeping up is even harder (Awada et al., 2018). As a result, children are either placed in a lower grade or they drop out of school (Karam et al., 2017). 22 percent of the Syrian refugee youth state that the difficulty of the Lebanese curriculum caused them to drop out while 8 percent state the language of instruction as reason for dropping out of school (UNESCO et al., 2014). 40 percent of enrolled Syrian refugee youth report understanding foreign languages as main difficulty in school (UNESCO et al., 2014).

Lack of Documentation: A lack of documentation and legal status prevents refugees from accessing education. The Lebanese government issues a variety of legal obstacles for Syrians to obtain residency permits, which lead to an increase of refugees without legal status (Basmeh & Zeitooneh, 2017). Many children are not registered as refugees with UNHCR because they cannot not afford the transport to the UN offices for their application or because they fear negative side-effects of their registration and mistrust the authorities (Karam Foundation, 2016). As undocumented refugees, they fear deportation or imprisonment which prevents them from moving freely. This impacts on their

educational access as many fear check points on their way to school. The lack of documents is also a problem when registering at public schools. Even though MEHE arranged that refugees are allowed to enrol regardless of their legal status, the awareness for this rule among school administrators is limited (El-Ghali et al., 2016). Among the 6-14 and the 15-17 year olds, the school prohibited them from enrolment in 7 percent of the cases (UNHCR et al., 2015). When attending school, a lack of documents is also a barrier when refugees try to sit for exams, as many schools require certificates of the previous class level to enter the exam or to receive certification after the exam even though the restriction is repealed by MEHE. As many refugees do not possess their Syrian documents and cannot access them without returning to Syria, they cannot enter higher classes (UNHCR et al., 2016). Despite how prevalent the issue is in all reports, it is not possible to find data on how many children are affected by the lack of documentation and legal status, probably as this topic is a very sensitive issue.

**Discrimination and Violence:** Hostilities from teachers and peers that lead to bullying and exclusion prevent Syrian children from accessing education (Visconti & Gal, 2018b). Parents raise security concerns when their children face corporal punishment, abuse and violence in classrooms or on their way to school and withdraw them from schools (El-Ghali et al., 2016). This is especially the case for girls when they or their parents fear sexual violence within or on the way to school, relating back to the aforementioned negative coping mechanism of early marriage. In 2012, the incidence of violence in the classrooms was at 70.4 percent (MEHE, 2014). Further statements on the prevalence of violence in classrooms are made in UNHCR et al. (2016), but there is no precise data to be found how many children are out of school because of fearing discrimination and violence.

Besides the aforementioned supply barrier to education, 6 percent of the children report that they are not enrolled because there is no school available in their area or because the schools are overcrowded and do not accept more students (UNHCR et al., 2015).

The combination of the public schools not being able to scale up their provision fast enough together with the lacking quality and the high drop-out rates indicate a need for after-school or out-of-school learning programmes that support Syrian children. Many Syrian and Lebanese community-based organizations have emerged to fill the capacity gaps in the formal school system by providing complementary or alternative informal education programmes. However, their performance is difficult to evaluate.

Section 4.1 drew on the historical background of Lebanon that was characterized by various conflicts, sectarian tension and foreign influence as described in the annex. Section 4.1.1 outlined the unprecedented influx of Syrian refugees in the aftermath of the Syrian war into Lebanon and section 4.1.2 described the economic, social and political impact of the Syrian refugee crisis on Lebanon, presenting how the previously existing structural problems and capacity gaps were exacerbated by the population increase. Section 4.1.3 illustrated the education sector prior to the refugee crisis and outlined

the existing capacity gaps. Section 4.1.4 showed how the difficulties in the education sector aggravated in the wake of the refugee crisis and how these problems are sought to be overcome with the emergency approach RACE I that was scaled up to the long-term strategy RACE II. Lastly, section 4.1.5 answered the first sub-research question by presenting the factors that constrain the access to education for Syrian refugees in Lebanon.

The education situation in Lebanon is worthwhile to be examined in the light of the theoretical framework of this thesis. It constitutes a service delivery in a fragile context, which according to OECD (2008) is disrupted by three factors. Firstly, the service delivery is disrupted by a lack of government capacity and/or willingness, the breakdown of social order through conflict as well as through corruption and private capture. Secondly, service delivery in fragile environments may be disrupted by the absence of a national policy framework that leads to the fragmentation of the action. Thirdly, the prioritisation of immediate service delivery over long-term capacity development may be unsustainable.

Regarding the first problem, Lebanon's struggling economy and the sectarian divide in the political-administrative system lead to a sincere limitation of the state capacity in delivering services and making decisions on the refugee situation. The sectarian organization of public life, from the organization of the political system to the provision of education, also impacts on the allocation of resources, which encourages favouritism along ethnic lines and thus private capture of public goods (Turner, 2015). These capacity gaps are accompanied by upsurges of conflicts with Syria, Israel, Saudi Arabia and Iran, and between the ethnic communities, which disrupts the social order and leads to intercommunal tensions. The Syrian refugee crisis aggravates this situation and fuels tensions between the Syrian refugees and the Lebanese host community, having the potential for violent clashes and a social breakdown.

Concerning the second problem, the absence of a clear response strategy of the Lebanese government to the Syrian refugee crisis and its late willingness to cooperate with UN agencies to develop the LCRP as framework disrupted the service delivery in Lebanon. Already before the refugee crisis the education delivery consisted of a fragment of public, subsidised and private schools whose capacity did not meet the demands. The Syrian refugee crisis exacerbated the fragmentation of education delivery under which the quality of education suffered for the host community and the refugees. The emergence of the RACE I and II plan as policy framework to tackle the issue came late, underfinanced and with few mechanisms to measure the progress (El-Ghali et al., 2016).

Referring to the third problem, the cooperation of Lebanon with the UN agencies can be viewed in two ways. The initial lead of the refugee response through the UN agencies followed an emergency approach of urgent service delivery as humanitarian support to the Syrian refugees and the Lebanese host community. In the education sector, the UN agencies supported Syrian refugee children who lack access to their national education system due to human-made crisis by providing supplemental education programmes in line with the emergency education concept of the Inter-Agency Network for Education in Emergencies (2009). The later formulation of the 3RP and the joint LCRP framework rather followed a long-term development approach to not undermine capacity development for the sake of urgent service delivery. This was important for the coordination between the local authorities and the international actors active in Lebanon to avoid OECD's (2008) second problem of fragmentation.

Therefore, the service delivery and especially the education delivery in Lebanon is rather disrupted by the first two OECD (2008) problems than by the third one. Regarding the state fragility definition of lacking the capacity and/or willingness to execute core government functions, this assessment indicates that Lebanon lacks the capacity to execute its socioeconomic governance of service delivery but shows willingness to some extent by cooperating on important framework policies (Hirschmann, 2016; OECD, 2008). However, this willingness is yet again limited by the sectarian divide. Moreover, this assessment underlines Brinkerhoff's (2010) argument that capacity gaps in the public service delivery are filled by non-state actors. In this case, the capacity gaps prior to the Syrian refugee crisis were filled by education provision through private or faith-based organizations and after the Syrian refugee crisis through a combination of international (UN agencies) and non-state actors. What initiatives these non-state actors take in the response to the education crisis is subject of the second sub-research question.

## 4.2 What are the Current Initiatives Taken by National and International NGOs Regarding Supporting Education Among Syrian Refugees in Lebanon?

The second sub-research question asks *"What are the current initiatives taken by NGOs and INGOs regarding supporting education among Syrian refugees in Lebanon?"*. In order to answer this question, section 4.2.1 provides an overview on NGO and INGO education provision in Lebanon. Section 4.2.2 introduces four education projects in detail and analyses to what extent they address the barriers to education previously identified, thus providing the answer to the second sub-research question.

#### 4.2.1 An Overview of Education Activities by NGOs and INGOs

In a fragile state such as Lebanon that faces a protracted refugee crisis, it is important to have alternative and complementary education initiatives for children who cannot access the formal school system. Therefore, the NGO sector is besides the MEHE the main provider of education services for Syrian refugees (El-Ghali et al., 2016). NGOs offer remedial classes to children who attend public schools, accelerated learning programmes to enable the integration in public schools, as well as literacy and numeracy skills for children who never attended school (El-Ghali et al., 2016). They can be distinguished between NGOs and INGOs.

In Lebanon, 47 INGOs coordinate their work in the Lebanon Humanitarian INGO Forum (LHIF) of which 13 are official partners of UNHCR Lebanon (UNHCR Lebanon, 2018). Some of these
partner INGOs are generally not active in the education sector such as the International Medical Corps, Medair, Interesos, International Relief and Development, Terre des hommes or the International Committee for the Development of Peoples (Interesos, 2018). Other INGO partners run education projects in other countries but focus on different sectors in Lebanon (e.g. Danish Refugee Council, Premiere Urgence International, Care, Plan, Oxfam and World Vision) (Danish Refugee Council, 2014; Interesos, 2018; Premiere Urgence International, 2018; UNHCR Lebanon, 2018; World Vision, 2017). The INGOs that are official partners of UNHCR Lebanon and are active in the education sector are Concern Worldwide, the International Rescue Committee and Save the Children.

Concern Worldwide started an education programme in 2014 that consists of literacy and numeracy classes for the integration into formal schooling and homework and retention support for enrolled students. The education programme also includes a psychosocial support component to mitigate the negative impact of violence and displacement. Concern Worldwide raises awareness among parents and trains teachers in first aid and child protection (Concern Worldwide, 2018). The International Rescue Committee (IRC) operated in the education sector through community-based NFE and basic numeracy and literacy programmes from 2012 to 2016. After the MEHE restricted humanitarian organizations from providing full NFE programmes outside of public schools without accreditation, the IRC shifted its focus in 2015 to early childhood education and remedial support to children already enrolled in public schools. The early childhood education programme covers literacy, numeracy and social and emotional skills (International Rescue Committee [IRC], 2017; United States Department of State, 2017). Save the Children offers remedial education, NFE programmes with homework support and basic literacy and numeracy classes as well as early childhood care across Lebanon (Save the Children Lebanon, 2018).

Apart from the official partner INGOs of UNHCR, other INGOs are engaged in the education provision in Lebanon. One of them is the Norwegian Refugee Council (NRC) which generally supports people affected by displacement and has been working in Lebanon since 2006. In the education sector, the NRC supports access and retention to public schools and creates NFE opportunities as well as vocational and language classes for out-of-school children. Apart from that it supports the UNRWA's schools for Palestinian refugees, trains teachers and rehabilitates public schools (NRC, 2018).

Aside from these INGOs, UNHCR Lebanon has 13 NGO partners, of which the Amel Association International, the Makhzoumi Foundation, the René Moawad Foundation and UTOPIA operate in the education sector (UNHCR Lebanon, 2018). The Amel Association International is active in Lebanon since 1979 and provides a NFE education programme with homework support, remedial classes and accelerated learning that is offered in its own Amel centres since 2017. It also targets children in informal camps and shelters through Mobile Education Units. The Amel Bus operates in five informal settlements on a weekly basis and offers retention support through homework and remedial sessions,

psychosocial and recreational activities and awareness sessions for parents (Amel Association International, 2017). The Makhzoumi Foundation offers a vocational education project for the youth in marginalised areas that trains them in digital literacy and English skills (Makhzoumi Foundation, 2018). The René Moawad Foundation provides education projects that focus on soft skills and hard skills, employability programmes that promote literacy and numeracy, vocational training, apprenticeships, life skills lessons and awareness sessions for parents (René Moawad Foundation, 2018). UTOPIA offers English language courses and recreational awareness camps on pluralism, tolerance and mutual respect (UTOPIA, 2018).

Besides the partners of UNHCR, other Lebanese NGOs are engaged in the context of education. One of them is Sonbola, an NGO founded in direct response to the Syrian refugee crisis in 2014 and operating in informal settlements in the Bekaa Valley. It delivers the Tamkeen NFE programme in its own learning centre which provides remedial education in Arabic, English, science and math. It also promotes computer and robotic education, psycho-social support through art and music as well as citizenship education through debate clubs. It uses blended learning approaches with ICT skills and immersive, interactive and experimental learning. Additionally, Sonbola runs the project Taleem that supports formal education's enrolment through free transport and awareness campaigns with parents and communities (Sonbola, 2018). Basmeh & Zeitooneh, is another example of an NGO that was founded as reaction to the Syrian refugee crisis in 2014. It runs a NFE project for students whose education has been disrupted by conflict or who cannot enrol in public schools. In three community centres the programme teaches the Lebanese curriculum with English, French, Arabic, science and math. Alongside of the NFE programme Basmeh & Zeitooneh offers a peace education programme with recreational and creative activities for the promotion of psychosocial well-being (Basmeh & Zeitooneh, 2017, 2018).

## 4.2.2 Answer to the Second Sub-Research Question: The Current Initiatives Taken by National and International NGOs Regarding Supporting Education Among Syrian Refugees in Lebanon

The aforementioned examples of NGO and INGOs education provision illustrate how diverse and multifaceted non-state education delivery in Lebanon is. Yet it is paramount to examine whether the diversity of approaches contributes to the education access of Syrian refugee children by targeting the critical factors that prevent them from following their education. These factors consist of the cost of education, negative coping mechanisms (e.g. child labour, child marriage), psychological stress and trauma, and being over age on the demand side. On the supply side, they consist of a lack of language skills or the difficulty of the curriculum, a lack of documentation, and discrimination and violence. In the following section, four education projects are introduced in detail in order to follow up on this question.

# Project 1: The Back-to-School Campaign by MEHE, UNHCR, UNICEF and Partner NGOs

#### **Description**

UNHCR has played an important role in ensuring access to education for Syrian refugee children. One of the initiatives taken in this respect is the *Back-to-School* campaign which operates before the beginning of each new school year. It is run, under the umbrella of MEHE, by UNHCR, UNICEF, UNESCO, the World Bank and more than 100 local NGO and INGO partners across the country and consists of a mass media campaign at the national level, an outreach and mobilization initiative at the community level, case management at the family level and preparation at school administrator level (MEHE, 2016). The information campaign is broadcasted on a multitude of information channels such as TV, radio, posters, information sessions, flyers, direct text messages, social media and an information hotline (UNHCR Lebanon, 2016).

The outreach campaign informs about the importance of education, enrolment opportunities and support mechanisms for already enrolled children (UNHCR Lebanon, 2016; UNHCR, 2017). Regarding enrolment opportunities, an important information for many parents is the partial waiving of documentation for enrolment. In cases in which some schools express additional requirements such as vaccination booklets UNHCR installs itself as communication partner that informs school administrators about legislative changes to ease the enrolment process (UNHCR Lebanon, 2016). Regarding support mechanisms, information on funding opportunities is crucial. Children who are enrolled in public schools through the *Back-to-School* initiative receive free text books and stationery. Moreover, for non-Lebanese children the tuition fees are waived and for Lebanese children the Parent Council fees at school (US\$60) are paid (UNHCR, 2017). Besides that, the most vulnerable refugees and those living more than 2.5 kilometres away from the nearest school can apply for additional cash support for transportation.

The campaign focuses on identifying and addressing reasons that hinder children from attending school and seeks community-based solutions for these issues. The information is gathered and distributed in multiple ways. The campaign trains refugees with a background in education, e.g. former teachers in Syria, as specialized Outreach Volunteers. The Outreach Volunteers organize awareness and information sessions for students and parents on what assistance they can receive and refer families in need to the respective service provider. Community Awareness Sessions are set up to identify problems through focus group discussions and to inform families about education opportunities in their houses, community centres, informal settlements and towns. Other outreach components are Parent Community Groups that involve parents and provide them with an opportunity to work on solutions for common problems together. For instance, in areas without public transportation to and from school, parents can set up transportation plans in which an adult accompanies the children. The precise design of the

outreach initiatives depends on the local partners and the community. One of the community outreach projects takes place in cooperation with the Lebanese scouts and is called the *School on Wheels*. It enables children to learn and play in a mobile classroom unit while their parents are informed about enrolment opportunities (UNHCR Lebanon, 2016). Another form of outreach is a cooperation with the Swedish Clowns without Borders that trains the outreach staff on trauma reduction methodologies in the *Learning through Laughter* approach (UNHCR Lebanon, 2016).

For the children that are already enrolled in an education programme, the parents are encouraged to establish Homework Support Groups. Doing homework together and under guidance motivates children to stay in school and offers a safe environment for questions. The Homework Support Groups take place in homes, tents, shelters and community centres and are led by Outreach Volunteers or parents to sustain the retention of children in education programmes. In addition, the *Back-to-School* campaign informs about opportunities for vocational education and higher education. It refers to one of 134 MEHE-accredited vocational education centres that teach practical skills and raise awareness for the DAFI scholarship programme that enables obtaining a university degree. Both initiatives are supported by UNHCR. Moreover, UNHCR initiates a pilot project in southern Lebanon for children with specific needs.

#### **Barriers**

Referring to the supply- and demand-side barriers to education, the Back-to-School campaign does not provide an education programme itself that seeks to overcome these barriers. Instead, it reaches out to communities through a multitude of channels and with a variety of local partners and approaches to identify barriers to education that have not been addressed and to inform about opportunities that are already in place to reduce barriers. Concerning the demand-side barriers to education, the Back-to-School campaign tackles the cost barrier of education by informing about the current regulation of free public education including free text books and stationery and by explaining the different funding opportunities that exist additionally. The funding opportunities include the cash support programmes for transportation in some cases or scholarships such as the DAFI programme. The financial burden on some families that urges children to work is at the core of the community-based outreach approach. It involves the parents and the community through Outreach Volunteers, Community Awareness Sessions and Parents Community Groups and seeks to inform them about the importance of education for their children's future. With an awareness for the importance of education in combination with the funding opportunities, the Back-to-School campaign contributes to a prioritization of education and thus positively impacts on negative coping mechanisms. The vocational education opportunities are an option to gain practical skills and execute more qualified positions in the job market for children who seek to work sooner. Regarding the barrier of psychological stress and trauma, the Outreach Volunteers are trained in initiatives such as the Learning through Laughter approach that equips them with trauma sensitive skills. Beyond this approach, no information is provided in the project report regarding alleviating psychological stress and trauma. Neither does the report provide insights into ways in which the over age children are being addressed. Presumably, the parents of these children belong to the families with specific needs that are referred to the support hotline or to one of the Basic Literacy and Numeracy Programmes that prepare children for the Accelerated Learning Programme and consequently for formal education in the long run.

In respect to the supply-side barriers to education, the *Back-to-School* campaign informs about additional language support classes and remedial support. The Homework Support Groups help children to keep up with the rest of the class. Concerning the documentation barrier, the campaign shares knowledge regarding enrolment procedures, required documentation and legal requirements and seeks to find a solution for every family. In cases in which school administrators refuse to enrol the children, their families can also speak up in the community outreach sessions or the support hotline. With regard to the barrier of discrimination and violence, the *Back-to-School* campaign's community outreach targets all children of school age and informs about funding opportunities for both, Syrian and Lebanese children. The Outreach Volunteers also have the responsibility to provide personalized follow up on refugee students on issues such as bullying, violence and discrimination (MEHE, 2016). In addition, the Parent Community Groups are instrumental for managing the relationships between the Lebanese and Syrian parents as well as for tempering tensions between the parents and the school administration and teachers. It is therefore a tool for addressing tensions that arise through violence or discrimination.

## Project 2: Learning in a Healing Classroom Programme by the International Rescue Committee

#### **Description**

During the school year 2016-2017, the International Rescue Committee (IRC) delivered the *Learning in a Healing Classroom (Healing Classrooms)* programme (IRC & Global Ties for Children at New York University [TIES/NYU], 2017b). It is designed as an after-school programme to support Syrian refugee children's learning outcomes and retention in Lebanese public schools. It is coordinated with the *Back-to-School* campaign by MEHE and is in accordance with MEHE's RACE plan and RACE II's Non-Formal Education Framework.

The *Healing Classrooms* programme is delivered by 145 IRC-trained teachers in 87 community sites in Lebanon's Bekaa and Akkar regions. The programme sites are located close to the homes or schools of the children. 4300 Syrian refugee children who are enrolled in Lebanese public schools receive eight hours per week of Arabic, second language (English or French) and math instruction. The programme is delivered in two 16-week cycles and there are three versions of the programme available. All of them implement a social-emotional learning (SEL) approach. Social-emotional learning affects

three realms of child development. Emotional processes such as recognition or empathy, social and interpersonal processes such as interpreting behaviour and clear communication, and cognitive processes such as working memory and attention control (IRC & TIES/NYU, 2017a).

The first version *Healing Classrooms Basic (HC Basic)* provides a five-day training to teachers to integrate SEL principles of classroom management, critical thinking and positive pedagogy into their teaching practices in Arabic, second language, literacy and numeracy. The second version *Healing Classrooms* + *Targeted SEL (HC* + *Targeted SEL Mindfulness)* includes additional daily mindfulness exercises in between subject-matter transitions to buffer psychological stress. These consist of breathing exercises or focusing on the present moment. The third version of the *Healing Classrooms* + *Targeted SEL Brain Games)* programme includes brain games based on movements and playfulness to build cognitive control in between subject-matter transitions. They are designed to support the working memory and flexibility of attention as well as overall executive functioning skills. The teachers of the *HC* + *Targeted SEL* versions *Mindfulness* and *Brain Games* receive an additional pre-service training (IRC & TIES/NYU, 2017b).

#### **Barriers**

Regarding the programme's ability to support education among Syrian refugees, the *Healing Classrooms* programme only reaches Syrian refugee children that are already enrolled in Lebanese public schools in the first place. However, it addresses the barriers why children drop out of the formal system and seeks to bolster the refugee children's ability to succeed in the formal system. Concerning the demand-side barriers to education, the programme sites close to the school or the home of the children reduce the indirect education costs for transportation and the insecurity on the way to the programme. The HC + Targeted SEL programmes additionally address the barrier of psychological stress and trauma with their mindfulness exercises and the brain game. Since the targeted children are already enrolled in the formal educations system, there are no barriers regarding the age-limit to be expected. However, the reports do not provide specific information on this topic.

With regard to the supply-side barriers to education, all three versions of the programme provide additional language instruction and math courses which address the language and curriculum barrier in Lebanese public schools in which math is taught in French or English. The reports do not provide information regarding the documentation that is required to participate in the programme. However, since the children have to be enrolled in the formal system, the entry into the after-school programme is supposedly not a problem. Regarding the barrier of discrimination and violence, the programme components do not specifically target the issue, but the teachers are taught in positive pedagogy which might contribute to a reduction of violent or discouraging teaching methods. The joint practice of the mindfulness components and brain games might contribute to a better classroom feeling. However, since the IRC programmes only target Syrian refugee children, their Lebanese peers might feel an aversion as they cannot benefit from the extra support.

# Project 3: Non-Formal Education Programming by the Norwegian Refugee Council **Description**

The Norwegian Refugee Council (NRC) provided NFE programmes to Syrian refugee children and vulnerable Lebanese children from 2013 to 2016 (NRC, 2017). Around 80 percent of the targeted children are Syrian refugees. Because the needs and the backgrounds of these vulnerable communities are diverse, the education programmes seek to be adjustable and flexible to these realities. Therefore, the NRC's education approach consists of three phases. It develops from a purely emergency-focused approach to a more sustainable non-formal programme and into a support programme that facilitates access to public schooling.

The first phase of the approach is an education in emergency programme for newcomers and out-of-school children. It provides an own *Child Education Pack* curriculum of three months that covers basic literacy and numeracy skills as well as psycho-social and recreational activities such as storytelling, sports, games and breathing and relaxation exercises. The programme aims at providing a sense of normalcy, to support children to cope with trauma and to teach basic educational skills. It is provided for up to one year and delivered in informal tented settlements or community centres (NRC, 2017).

The second phase of the approach consists of structured NFE programmes that aim at integrating the children into the certified public education system. They are based on the national Lebanese curriculum and are in line with MEHE's RACE plan and RACE II's Non-Formal Education Framework. The out-of-school children who are attending the structured NFE programmes are prepared for the formal system and are transferred either to the regular formal school system or the MEHE implemented Accelerated Learning Programme. Therefore, the NRC programmes are implemented in the same three-month cycles than the official MEHE Accelerated Learning Programme. One of them is the community based early childhood *School Readiness* education programme for pre-school children. The second one is the *Basic Literacy and Numeracy* programme for out-of-school children. They are delivered in community centres or NRC-owned learning centres because MEHE limits the access to public schools since it standardizes and centralizes NFE programming (NRC, 2017).

The third phase of the approach are support programmes to formal education to prevent dropout and to ensure retention in the formal school system. The remedial *Learning Support* programme targets public second shift schools as they specifically provide education for Syrian refugee children. Because of the double-shift system, these schools lack the capacity for additional learning support classes, wherefore the NRC *Learning Support* only operates during the summer break inside these schools. Children who perform poorly are referred to the NRC support classes by headteachers. In addition, the NRC offers the *Homework Support* programme (NRC, 2017).

In all three phases, the NRC deploys a holistic approach. It provides learning and academic skills with psycho-social and recreational support and life skills training. The NRC involves parents through *Parent Community Groups* and awareness and information campaigns on child rights, hygiene and positive parenting. It participates in *Community Outreach Campaigns* such as the MEHE *Back-to-School* campaign and encourages school enrolment through household visits, community awareness sessions and follow-up phone calls. The NRC trains and supports teachers and school personnel with child-centred inclusive teaching methodologies, multi-level teaching, positive discipline and the integration of light psycho-social activities. The latter are implemented throughout all three phases with NRC's psycho-social support programme *Better Learning*. In addition, the NRC creates a safe learning environment by rehabilitating schools, repairing learning spaces and distributing learning materials (NRC, 2017).

#### **Barriers**

Regarding the demand-side barriers to education, all NRC programmes run for free and do not involve direct costs for tuition. The programmes operate in community centres in the South and North and in informal tented settlements in the Bekaa. The proximity to the community enables assess and contributes to the reduction of indirect costs as no transportation is needed. However, children who do not live close to these centres are not specifically targeted through e.g. school busses. The distribution of learning material also lowers the indirect costs of education. Outreach campaigns and the strong involvement of the parents and the community contribute to the awareness for the importance of education. This might lead to a prioritization of education so that children are not send to work or married early. Another focus of the NRC programmes lies on the barrier of psychological stress and trauma as the NRC implements the psychological support programme *Better Learning* throughout its three phases. Especially in the emergency education phase, different psycho-social activities and relaxation exercises target the psychological well-being of the children. Also, the teachers are specifically trained for psycho-social activities. Regarding the barrier of being over age, the projects report does not provide information how the children are categorised in age groups and thus possibly excluded.

With respect to the supply-side barriers to education, the language and curriculum barrier is targeted. The emergency education programme extra focuses on the most important basic life skills that are relevant in crisis situations. The other two phases of the NRC programmes are closely aligned with the official Lebanese curriculum. It is not laid out how many hours of which subject are taught and if there is a specific focus on languages. But since the goal of the NFE programmes is to integrate the children into the formal system, they supposedly focus on the subjects in which the children face the biggest challenges, namely math and foreign languages. As even the first grade of the formal school

system as well as the MEHE Accelerated Learning Programme presume basic literacy and numeracy skills, the focus on literacy and numeracy in the NRC Basic Literacy and Numeracy programme targets this barrier to education. The Learning and Homework Support programmes focus precisely on the areas in which the children struggle to sustain their retention. Information how in detail the Basic Literacy and Numeracy or the Learning Support programme communicate and teach these skills is not provided in the project report. Neither does the report provide insights into the regulations of the NRC regarding the documentation that is needed to enrol in the NFE programmes. If the same documentation is needed that prevents children from enrolling in the formal school system, these children continue to be excluded. On another note, the NRC education programmes target Syrian refugee children and vulnerable Lebanese children alike. Thus, they do not perpetuate or fuel resentments that marginalized Lebanese communities are left behind while Syrian refugee children receive support. This can have a positive effect on the barrier of discrimination and violence. However, since 80 percent of the children are Syrian refugees, this effect might be rather small. The aim of integrating the refugees in the Lebanese public formal system increases the possibilities of interaction with Lebanese peers, but the second-shift approach prevents a full integration. In addition, the NRC teacher training addresses the discrimination and violence barrier by promoting an inclusive classroom atmosphere and by showing methods of positive discipline as alternative to (corporal) punishments.

## Project 4: The Rainbow of Hope Project by Search for Common Ground <u>Description</u>

The *Rainbow of Hope* project of Search for Common Ground (SFCG) was funded by the U.S. Embassy in Beirut and implemented by local NGOs in 25 locations across Lebanon from November 2014 to July 2015 (Search for Common Ground [SFCG], 2015). It aims at overcoming tensions between the Syrian refugees and the Lebanese host community through NFE programming that promotes conflict transformation. For the duration of six months, 625 Lebanese, Syrian and Palestinian children between six and 11 years are supported with English classes. They also receive psycho-social and recreational activities together which an additional 625 children.

The English classes are delivered from Monday to Friday in the afternoons for one hour each. The classes teach basic and intermediate English language skills through language games, music and storytelling. 25 English short stories target conflict transformation topics such as empathy, respect and mutual understanding and seek to promote community building. The classes offer discussions on these topics in Arabic and incorporate English for resolving conflicts in a peaceful way. The psycho-social and recreational sessions are delivered by two animators per club for four hours on a weekend day in the locations were also the English classes take place. At the end of the project, three final events are organized as an opportunity to bring the parents and students together and present the progress that has been reached (SFCG, 2015).

In order to implement the project, SFCG recruits 25 teachers and 50 animators and trains them for the programme components. The teachers receive five days of trainings on English teaching methods and classroom management as well as tools to integrate conflict transformation skills into the English clubs. They also write their book profiles and prepare their weekly lesson plan. The teachers receive one Refresher Day in which they get trained in child protection and new education tools. The animators receive three days of training on how to include conflict transformation skills into recreational activities and how to deal with conflict among children. Also, the animators receive one Refresher Day that focuses on creative tools for child protection and overcoming behavioural problems of children. Throughout the project the teachers and animators also receive coaching (SFCG, 2015).

#### **Barriers**

Referring to the demand-side barriers, the project focuses on out of school children in the most vulnerable communities and thus targets children that are otherwise excluded from education. Since the project is offered for free and in close proximity to the children's home, the barrier of education cost is tackled. The children are not wearing uniforms on the published pictures which indicates that also these indirect costs are avoided. In terms of negative coping mechanisms, the project report states that the programme aims at combatting child labour by providing the incentive of English language learning for Syrian refugee families to keep their children in education. Since no other incentives are provided, the contribution of the project in this realm is not evident. The psycho-social activities for the children on the weekend target the barrier of psychological stress and trauma. The focus on children from six to 11 years includes a wide age range wherefore also older children with no prior exposure to English can attend the classes, possibly contributing that children that are over age for their grade level in formal education can participate in this project.

Regarding the supply-side barriers, the project focuses extensively on the provision of English language skills which corresponds with the language and curriculum barrier to education. The teacher training promotes an understanding for different learning styles which prevents teaching on the lowest common denominator given the age-range from six to 11 years and the accompanying variations in comprehension level. However, the programme's focus on English does not address the overall literacy and numeracy of the children. Concerning the barrier of lacking documentation, the project report does not provide insights to what extent documentation is required for the enrolment in the programme. The project's component on conflict transformation skills inhibits a strong focus on overcoming the barrier of discrimination and violence. The whole project is directed towards community building, overcoming prejudices, fostering mutual understanding and peaceful conflict resolution between Syrian, Lebanese and Palestinian children, thus impacting on bullying and other forms of discrimination and violence in the classroom. The project report mentions that the project is implemented where SFCG already runs other social cohesion projects due to a high concentration of Syrian refugees. Therefore, the *Rainbow of* 

*Hope* project carries the social cohesion message into the classrooms. However, there is hardly any integration of the wider social cohesion projects with the *Rainbow of Hope* project and the only community or family involvement in the project takes place in the final events.

The second sub-research question presented an overview on NGO and INGO education projects in Lebanon and provided an in-depth analysis of four education projects. Project one profoundly targets the barriers cost of education, lack of language skills or curriculum and lack of documentation. Project two is particularly strong in targeting the barriers psychological stress and trauma, the lack of language skills or curriculum and moderately addresses discrimination and violence. Project three strongly impacts on psychological stress and trauma, the lack of language skills or curriculum, and discrimination and violence, and moderately targets the lack of documentation and cost-related barriers. Project four targets the barriers cost of education, psychological stress and trauma, being over age, lack of language skills (not curriculum), lack of documentation, and discrimination and violence, but is only punctual in scope.

# 4.3 What is the Documented Effectiveness of the Selected Education Projects in Reducing the Main Barriers to Access to Education for Syrian Refugees?

The third sub-research question asks "*What is the documented effectiveness of the selected education projects in reducing the main barriers to access to education for Syrian refugees*?". In order to answer this question, the sub-chapter 4.3 elaborates on the monitoring mechanisms that are present in the four education projects and presents the goal achievement of the respective projects.

# Project 1: The Back-to-School Campaign by MEHE, UNHCR, UNICEF and Partner NGOs

#### **Monitoring Instruments**

The *Back-to-School* campaign seeks to support children's access to education through awareness raising and spreading information on enrolment opportunities and support mechanisms. The following monitoring indicators are in place in order to track the progress of the programme.

In general, UNHCR as well as UNICEF employ a variety of monitoring tools at the macro and micro levels to ensure a high quality of their programming for children. At the micro level, they conduct monitoring through field monitoring, partner reports and Third Party Monitoring to track outputs and activities and to ensure the quality of the respective programme implementation (United Nations Children's Fund Lebanon [UNICEF Lebanon], 2016). At the macro level, the outcomes of UNHCR's or UNICEF's contribution are monitored through surveys and evaluations (UNICEF Lebanon, 2016). For the *Back-to-School* campaign, these monitoring instruments are also deployed. However, the programme report does not provide insights into the precise indicators. It is known that the Outreach

Volunteers and partner NGOs who inform the communities about the education opportunities gather the concerns of the communities expressed in the household visits and other forms of collaboration and report back to UNHCR (UNHCR Lebanon, 2016). Furthermore, direct feedback from beneficiaries through regular focus group discussions, community groups and text messaging is taken into account (UNHCR Lebanon, 2016). Based on this feedback, the barriers to education can be identified and addressed through legislation and programming. What exact information of the household visits is collected is not known.

#### **Goal Achievement**

These monitoring instruments enable to identify to what extent the goal of raising awareness on the importance of education, enrolment opportunities and support mechanisms and consequently improving children's access to education is achieved.

In the academic year 2015-2016, 354.936 children are enrolled in Lebanese public schools of which 157.984 are Syrian refugee children (UNHCR Lebanon, 2016). This is double the enrolment rate of the 2013-2014 academic year thanks to the introduction of the double shift system and the *Back-to-School* campaign (UNHCR Lebanon, 2016). Furthermore, 29.000 children access NFE programmes such as MEHE's Early Childhood Education, Basic Literacy and Numeracy and the Accelerated Learning Programme (UNHCR, 2017). However, of the 488.000 school-aged Syrian refugee children, the target of enrolling 200.000 refugee children in formal education is not reached so that an estimated 180.000 Syrian refugee children in primary school age (6-14 years) remain out of school (UNICEF Lebanon, 2016; UNHCR Lebanon, 2016).

The *Back-to-School* campaign mobilises 109 Outreach Volunteers for the community-based approach and initiates 83 Parent Community Groups as key initiative to involve parents in their children's education. 101 Homework Support groups are established to sustain retention in schools. They are popular especially in the mornings before the second school shift starts. Moreover, 1.365 Community Awareness Sessions are held and they reach 24.451 parents. The information sessions lead to the enrolment of 1.033 students in vocational education and enable 131 students to receive a DAFI scholarship for Lebanese universities. Lastly, the outreach campaign identifies 170 children with special needs who receive glasses, hearing aids and wheel chairs (UNHCR Lebanon, 2016).

Project 2: Learning in a Healing Classroom Programme by the International Rescue Committee

#### **Monitoring Instruments**

The *Healing Classrooms* project aims at supporting Syrian refugee children that are enrolled in Lebanese public schools to succeed in the formal school system. In order to reach that goal, the following monitoring mechanisms are deployed.

The *HC Basic* version provides teachers with regular monitoring visits and peer-support through Teacher Learning Circles once a month (IRC & TIES/NYU, 2017b). The *HC* + *Targeted SEL* versions *Mindfulness* and *Brain Games* receive additional in-service monitoring to support their use of the SEL activities (IRC & TIES/NYU, 2017a, 2017b). Syrian refugee children who receive the *Healing Classrooms* programme (any of the versions) are compared to Syrian refugee children without access to this programme before the start of the programming, after the first 16-week cycle and after the second 16-week cycle. The impact evaluation deploys the Early Grade Reading Assessment (EGRA), the Early Grade Math Assessment (EGMA) as well as additional factor analysis to assess the literacy and numeracy skills (IRC & TIES/NYU, 2017a). The cognitive, social and emotional skills are monitored with a scenario-based measure in which children are asked to hypothetically respond to certain social situations (IRC & TIES/NYU, 2017b). Furthermore, 80 trained Lebanese enumerators assess the children's working memory, inhibitory control and executive functioning skills with a tablet-based research assessment with children, parents and teachers (IRC & TIES/NYU, 2017a). Moreover, qualitative interviews with students and parents are conducted in order to assess the recipients' perception of the experience (IRC & TIES/NYU, 2017a, 2017b).

#### **Goal Achievement**

With regard to the output of the *Healing Classrooms* programme, it is relevant whether and to what extent the goal of supporting Syrian refugee children that were enrolled in Lebanese public schools to succeed in the formal school system is achieved.

Regarding the access and retention in formal schooling, there is no data available how many Syrian refugee children drop out of the programme or out of the formal school system in comparison to children who do not have access to the programme. The data that is available shows that the students struggle to attend the retention support classes as they attended only 40 percent of the 51 retention classes. At the first assessment, one third of the Syrian refugee children cannot recognize one letter of Arabic, the language of elementary instruction in Syria and Lebanon. In the EGRA, 30 percent of the students score zero in all sub-tasks. In the subsequent mid-term and end assessments, the children with access to the programme have higher Arab reading and math skills compared to Syrian refugee children without access to the *Healing Classrooms* programme. However, the reports do not provide the information to what extent the results in e.g. the EGRA improve. The children who attend the *Healing Classrooms* programme are less likely to view interactions with peers as hostile in intent which is called the hostile attribution bias and results from bullying or violence. Moreover, the children recognize and

express their feelings, including the negative ones. However, there are no significant improvements of the children's mental health such as depression or anxiety (IRC & TIES/NYU, 2017a, 2017b).

### Project 3: Non-Formal Education Programming by the Norwegian Refugee Council <u>Monitoring Instruments</u>

Also, the three phases of the NRC's education programme require monitoring mechanisms in order to trace the process in reaching the goals. The goals are the provision of a sense of normalcy and the buffering of a crisis situation through emergency education, the integration of the children into the formal system through the pathway of NFE programmes, and the retention of the children in the formal system, preventing their drop-out through support programmes (NRC, 2017).

The monitoring of the NRC education programme operates on different levels. NRC keeps a child-level Excel database which includes the attendance, retention and completion rates of every child that is enrolled in the NRC programmes. In the second phase, children are tested in their physical, emotional and cognitive development before entering and after completing the NFE programmes. In the third phase, the NRC conducts a pre- and post-test with the children to measure their academic improvement and the effectiveness of the *Learning Support* programme. Across the phases, the lessons are assessed with structured observations to ensure the quality of the teaching, to provide feedback for the teachers and to decide if the teacher training needs to be adjusted (NRC, 2017).

In addition, the NRC conducts phone interviews with the parents of the children who attend NFE classes and are referred to public schools in the 2015/2016 academic year. In 2017, after almost two years, they are asked with a structured questionnaire whether their children are still enrolled. The interviews are also used to assess if there is a difference between children who attended the *Child Education Pack* (phase one) and the *Basic Literacy and Numeracy* classes (phase two). In addition, it is examined if the retention and transition rates of the children in the *Learning Support/Remedial* programme (phase three) improve (NRC, 2017).

Moreover, the NRC conducts spot-checks to control for the compliance with donor agreements. The NRC feeds these monitoring indicators into its (internal) global outline database to track monthly progress on the output and outcome level indicators and to measure them against the targets. Information on what exactly these indicators consist of is not provided in the report (NRC, 2017).

#### **Goal Achievement**

After pointing out the monitoring instruments that are in place to track the process in reaching the goal of the NRC education programme, the outputs of the three phase-approach are presented.

Between 2013 and 2016, NRC enrols 42.454 refugees and host community children (ca. 10.000 children per year) in NFE programmes (NRC, 2017). The outreach campaigns at the beginning of each academic year refer 22.390 children to formal public education (NRC, 2017). The assessment of the Basic Literacy and Numeracy programme (phase 2) in the south of Lebanon in 2016 shows that 76 percent of the children who attend the classes enrol in formal public education. Moreover, in the assessment of the Learning Support programme (phase 3) in the 2015/2016 academic year 88 percent of the children show improvements in Arabic, math and foreign languages (NRC, 2017). Out of these students, 28 percent improve their academic performance at the highest performance benchmark. With regard to the teacher training component, NRC reaches 1.300 teachers. When assessing them with the standardized quality lesson observations in the 2016/2017 academic year, 83 percent of the trained teachers are able to replicate the key methods of the training (NRC, 2017). Regarding the involvement of the community, NRC engages 11.000 parents in Parent Community Groups of which 15 percent actively organise activities such as a save access to the NRC learning centres (NRC, 2017). Moreover, the results of the telephone interviews in which the emergency approach (phase 1) and the non-formal learning programme Basic Literacy and Numeracy (phase 2) are compared show that the registration, retention and transition rates of the students in the Child Education Pack and the Basic Literacy and Numeracy programmes are similar (NRC, 2017). Even though the registration rates are still low, once the children are registered, the retention and transition rates are relatively high. The children who attend the NFE programme and additionally the Learning Support classes show the highest enrolment, retention and transition rates (NRC, 2017). However, the report does not provide more detailed data on the comparison between the programmes and the effectiveness of their combination. Comparative data from MEHE regarding the success rates is lacking, wherefore a comparison between children who attend NFE programmes and children who do not is not possible (NRC, 2017).

#### Project 4: The Rainbow of Hope Project by Search for Common Ground

#### **Monitoring Instruments**

In order to reach the goal of enhancing the English literacy and the conflict transformation skills through English classes and recreational activities, the *Rainbow of Hope* project deploys several monitoring instruments to track its process.

Firstly, the local NGOs reports to the SFCG on a monthly basis. With regard to the English classes, a variety of monitoring instruments are in place. On the teacher level, the SFCG distributes observation sheets to be filled out four times during the project (December 2014, January 2015, March 2015, May 2015) to examine the pedagogical and behavioural improvements (SFCG, 2015). The teachers are asked to track the attendance of students on a daily basis with an attendance sheet. In addition, they prepare and evaluate session plans for each short story. Moreover, the SFCG project coordinator and education advisor conduct field visits to the classes once a week and prepare a field visit

report. The classes are assessed in terms of lessons, design, students' progress, positive environment between students from different geographical and cultural backgrounds as well as the level of collaboration of the teachers to send their reports on time (SFCG, 2015). With regard to the recreational activities, the animators are also given observation and attendance sheets for each weekend and write a monthly report in order to track the behavioural improvement. In addition, the SFCG education advisor conduct interviews with a teacher and eight students of one of the project centres and further interviews with other students and their parents at the end of the project (SFCG, 2015). The SFCG project coordinator conducts phone interviews with ten animators at the end of the project. The final events are used to examine the children's interaction among each other. However, no information is provided to what extent and how the interactions during the events were assessed. Lastly, all of the aforementioned monitoring instruments were condensed in an Excel sheet (SFCG, 2015).

#### **Goal Achievement**

By deploying these monitoring mechanisms, the *Rainbow of Hope* project reaches the goal of enhancing the English literacy and the conflict transformation skills in the two areas of English classes and recreational activities to the following extent.

With regard to the English classes, the teachers' confidence in their capacity to teach the English curriculum improves from 63 percent to 96 percent. They most frequently use methods of vocabulary games, drawing or playing the main characters of the short stories, exploring moral lessons of the short stories and paraphrasing the stories in own words (SFCG, 2015). In the end, students are capable of linking the stories to their own life by telling similar stories. Each English class is supposed to cater an equal number of Syrian and Lebanese children. On average the classes are delivered to 42 percent Lebanese, 53 percent Syrian and 5 percent Palestinian students (SFCG, 2015). Of the 24 students per class, 16.8 students attend all classes. On average, the student's English language skills improve by 28.7 percent from initially 21.9 percent to 50.6 percent (SFCG, 2015). More specifically, 95 percent of the students can recognize, pronounce and use simple English words, 87 percent can describe pictures with English words, 70 percent comprehend the main ideas of the short stories and 35 percent can retell the short stories with the new vocabulary (SFCG, 2015). 14 percent of the students can express themselves in English and 12 percent can describe pictures using English sentences, which is the most advanced English skill (SFCG, 2015).

The English classes also impact on the children's conflict transformation skills. The children's behaviour improves on average by 26.7 percent from 27.6 percent to 54.3 percent. More specifically, 93 percent of the students positively interact with children from different nationalities and 82 percent approach others with positive intentions (SFCG, 2015). 61 percent listen to others even if they disagree and 29 percent respect and understand other persons' rights to learn and have fun (SFCG, 2015). The

students behaviour in conflict situations shifts from running away or fighting to resolving problems and using conflict management behaviour such as asking a teacher for conflict mediation (SFCG, 2015).

Concerning the recreational activities on the weekends, the animator's understanding of different animation techniques increases from 13 percent before the training to 76 percent after the training (SFCG, 2015). The project improves the animators' confidence to promote conflict transformation principles from 53 percent before the training to 87 percent after the training (SFCG, 2015). The extent to which they can promote these skills among children increases by +0.1 from 4.3 to 4.4. In the recreational activities an average of 39 children participate per weekend and centre, of which 12 children attend the English classes and the recreational activities (SFCG, 2015). 53.8 percent of the students are Syrian, 35.9 percent Lebanese and 10.3 percent Palestinian. The monthly reports of the animators shows that the quality of interactions among the children (on a scale from 1 with very negative interactions to 5 very positive interactions) increases by +0.8 in the Bekaa region, by +0.2 in the North and remained the same in the South (SFCG, 2015). It thus improves on average by +0.3. Furthermore, there is a -0.4 reduction of violent incidents, demonstrating the effect of the conflict transformation methodology (SFCG, 2015). Strong prejudices on both sides such as children not talking to each other shits from to more interaction. The verbal violence (talking down to others, bullying, racist behaviour, swearing) and physical violence (kicking, fighting, pushing) as well as gender-based discrimination (girls were told not to speak) are also improved. The animators find an improvement of +0.5 in the children's ability to positively deal with conflicts from 3.6 to 4.1 (on a scale from 1 not at all to 5 totally) (SFCG, 2015). The third component of the project are the final events. In those, 150 Lebanese and Syrian children and their families participate in the South, 175 in the North and 170 in Bekaa. (SFCG, 2015).

This is a remarkable result given that in the beginning prejudices such as "I don't want to sit next to a Syrian, my mom does not allow me sit next to them because they are not clean" or "Lebanese think they are better than us" prevailed (SFCG, 2015, p. 22). However, one education project is not enough to overcome the social tensions that exist in the vulnerable communities and a greater involvement of the parents would be needed so that the prejudices that are overcome in class are not reproduced at home. The same is true for discrimination of girls in the project. The project cannot overcome patriarchal structures, but it raises awareness and communicates respect and the right to learn of all children including girls and thus contributes to the reduction of violence as indicated above.

The third sub-research question focused on the effectiveness of the education projects by examining whether they monitor their progress and if the project output fulfils the targets that were set in the beginning. All four projects deployed reporting, monitoring visits and field observations as monitoring instruments. Project one additionally included direct feedback through focus group discussions; project two, three and four conducted pre- and post-intervention tests; and project three and

four interviewed participants, teachers or parents. Project one fulfilled its target of raising awareness and informing on enrolment opportunities and support mechanisms by reaching almost 25.000 parents but did not reach the target of enrolling 200.000 Syrian refugee children. Project two sought to support Syrian refugee children to succeed in the public school-system and reached the target of improving the Arab literacy and math skills and reducing hostile interactions, but it did not reach the goal of improving the children's mental health. Project three aimed at integrating children into the formal school system which was achieved in 76 percent of the children attending the NFE programmes, and contributing to their retention, which was achieved due to an 88 percent improvement in Arab literacy, math and foreign languages. It is unknown whether the aim of providing a sense of normalcy through the emergency programme was reached. Project four targeted to improve the English literacy and conflict transformation skills which was reached with a 28 percent increase of English literacy and a 26.7 percent improvement of children interacting peacefully.

### 5. Conclusion

To draw a picture on the education situation of Syrian refugees in Lebanon, the first sub-research question presented the intertwining between Lebanon's fragile service delivery and the impact of the Syrian refugee crisis by elaborating on the response to the refugee crisis, the socioeconomic repercussions of the crisis, the challenges of the education sector prior to the crisis and the adjustments that have been made to cater the increased service demand. Section 4.1.5 identified the constraints to education access for Syrian refugees and consequently answered the first sub-research question. To sum up, the cost of education, negative coping strategies (child labour and child marriage) and being over age are the main demand-side barriers to education while psychological stress and trauma are found to be widespread but without evidence on their impact on enrolment. A lack of language skills and the consequential difficulties in following the curriculum are the main supply-side barriers to education whereas a lack of documentation and discrimination and violence were found to be widespread but without data on the direct link to education. On a structural level, the barriers to education in Lebanon derive from education delivery being disrupted by capacity gaps and sectarian tensions that result in a lack of social cohesion and fragmented action as defined in OECD (2008). Based on these findings, the second sub-research question explored current initiatives by NGOs and INGOs regarding supporting education access for Syrian refugees. Section 4.2.1 found that the initiatives either focus on integration into the formal school system through NFE or retention in the formal school system through remedial and homework support. Most of them also incorporated psycho-social and recreational activities. Section 4.2.2 subsequently analysed the four education projects (1) *Back-to-School* campaign by MEHE, UNHCR, UNICEF and partner NGOs, (2) Learning in a Healing Classroom by the IRC, (3) Non-Formal Education Programming by the NRC, and (4) Rainbow of Hope by Search for Common Ground in terms of the barriers to education that they address. The third sub-research question analysed the effectiveness of these four education projects with effectiveness being defined as deploying monitoring instruments and reaching the set project targets. All four projects deployed a variety of monitoring instruments such as monitoring visits or pre- and post-intervention assessments. Project one and two reached their targeted objective partly, whereas project three and four achieved their goals fully.

Drawing on the answers of the three sub-research questions, the main research question "How and to what extent do selected non-state education projects address the factors inhibiting the access to education for Syrian refugees in Lebanon effectively?" is provided. Among the four education projects by non-state providers, the demand-side barrier cost of education was tackled only indirectly by waiving fees, providing materials or informing about funding possibilities (project one). There is limited evidence that the contribution of any of the four projects has been substantial regarding overcoming negative coping strategies such as child labour and child marriage. Awareness campaigns and the inclusion of parents and communities (project one and three) are not sufficient to tackle this issue. Instead, structural solutions that take into account the interrelatedness with sectors such as health and shelter are necessary for improving the dire living conditions of refugee families and alleviating the children's need to support their families financially. Awareness regarding the importance of psychological well-being and trauma-sensitivity is reflected in all four projects through training teachers (project one) or involving social-emotional and psycho-social activities (project two, three and four). The barrier of being over age is only substantially tackled by project four that includes out-of-school children of a wide age-range. The lack of language skills and the difficulty of the curriculum are addressed through distributing information on support opportunities (project one) or directly offering remedial support (project two and three), accredited NFE (project three), language courses (project two, three and four) or emergency education to out of school children (project three and four). The lack of documentation is, together with being over age, a barrier that is not sufficiently addressed, especially in projects that offer only remedial support as it excludes children that are not enrolled in the formal school system in the first place (project two). In this respect project three and four are more flexible by offering emergency education for out of school children. Further attention is also needed regarding discrimination and violence. Project two and three target this barrier only indirectly through mindfulness activities and teacher training on positive psychology or inclusive non-violent methodology. Only in project four overcoming discrimination and violence forms an integral part of the curriculum through the promotion of conflict transformation skills.

This multitude of education deliverers with varying goals can be interpreted with Matland's (1995) bottom-up approach in which the central actor MEHE devises the government programme RACE I and II on the macro-implementation level while NGOs and INGOs develop and implement their own programmes on the micro-implementation level. RACE consists of a clear goal, ensuring access to education, but allows for ambiguity in the choice of means. The high ambiguity leads to outcome variation from project to project and the low conflict enables a multitude of NGOs, INGOs and other non-state actors to participate in the education delivery, thus representing an experimental policy

implementation process (Matland, 1995). Project three's development from a purely emergency-focused response to a three-phase MEHE-accredited education approach is illustrative of experimental implementation in which the project starts with little knowledge and learns along the implementation process how to best respond to local needs (Matland, 1995; NRC, 2017). The project outcome is thus determined by the context in terms of actors involved at the local level and recourse available.

In stable contexts, seeing the government not as sole provider of services is a characteristic of NPM reflected in contracting out. This component was analysed in a fragile context through the thesis' focus on non-state education delivery in Lebanon, generating knowledge on service delivery in fragile states. It contributed to a greater understanding how non-state actors operate in cases of inadequate state service delivery and showed that the capacity gaps in the state service delivery do not result in a vacuum but are filled with more less formal mechanisms between state and non-state actors. The analysis of INGO education projects contributed to an understanding of the NGO activities in Lebanon, presenting that they form an indispensable pillar in the education sector. Identifying the factors than constrain Syrian refugee children from accessing education showed that all demand- and supply-side education barriers defined by Aydin and Kaya (2017), Bademci et al. (2016), Justino (2016), Karam et al. (2017), Kirk (2007), Sunny et al. (2017), Visconti and Gal (2018a) and Wofford and Tibi (2018) apply to the Lebanese context, thus validating their research findings. Examining the education barriers also provided insights into the hardships Syrian refugee children face and showed that despite the efforts a tremendous proportion of children remain out of school, thus underlining the importance of scaling up the efforts. The analysis of the extent to which the barriers to education are addressed by education projects identified effective projects that can be used for extracting best practices. It also identified projects that require readjustments to cater the needs of refugees effectively. Taking these findings into account in the planning process of future education projects might be beneficial to avoid repeating mistakes and reaching children that are now still excluded. Two of these mistakes are the sole focus on Syrian refugee children without addressing vulnerable Lebanese children and focusing on remedial education projects that target only children already enrolled in the public school-system. The outline of the education framework in which the NGOs work also showed that more coordination is necessary. Given that many children pass through different types of education programmes before finishing primary education, a joint accreditation system and increased recognition are needed.

Over the course of the research process the availability of empirical data on NGO and INGO education projects was limited wherefore the thesis only addressed four projects in depth. The researcher compensated this by providing an overview on smaller education projects. The publishing of reports in Arabic and French further limited the data pool. Given that not all Syrian refugees in Lebanon are registered, the researcher dealt with deviating enrolment and retention rates depending on the source and the absence of a central data system that includes information on NFE programming. Despite posing a barrier to education the issue of child marriage was not sufficiently addressed in the thesis due to a

lack of data. Further research on barriers to education could therefore deploy a gender-sensitive approach and distinguish between barriers for girls and boys. The strength of the in-depth information on Syrian refugees in Lebanon in turn creates the weakness of low generalisability for other country contexts. However, the neighbour countries that face a similar refugee situation such as Jordan can maybe derive information with caution for their NFE programming from Lebanon's case. Further research could therefore zoom out and look at the regional level or zoom in and distinguish specific barriers on the governorate level. To counter the data issues, future research should engage in the country through interviews and surveys and thus seek to close the data gap on NFE programming with a robust data system. Since censuses and other central data acquisition is a politically sensitive issue in Lebanon, researchers can maybe encourage progress where political actors cannot. Lastly, further research could examine the structural deficiencies in education delivery in Lebanon elaborated in section 2.2 and 4.1.2 that scattered NFE cannot compensate. In this respect, post-colonial research will not only apply a Westphalian sense of statehood to fragile settings but consider other institutions that inhibit authority and legitimacy in hybrid political orders (Peace Security and Development Network, 2009; Stel & van der Borgh, 2017). The provision of this data brings us one step closer to achieving RACE's goal of ensuring access to education for every child and overcoming the factors that constrain education delivery for Syrian refugees in Lebanon.

### **6** References

- Amel Association International. (2017). Mobile Education Units. Retrieved from http://amel.org/projects/mobile-education-units/
- Awada, G., Diab, H., & Faour, K. (2018). A call for curriculum reform to combat refugees crisis: the case of Lebanon. *The Curriculum Journal*, 29(1), 43–59. https://doi.org/10.1080/09585176.2017.1400450
- Aydin, H., & Kaya, Y. (2017). The educational needs of and barriers faced by Syrian refugee students in Turkey: a qualitative case study. *Intercultural Education*, 28(5), 456–473. https://doi.org/10.1080/14675986.2017.1336373
- Babbie, S. (2013). The Practice of Social Research. Retrieved from http://jsp.ruixing.cc/jpkc\_xingfa/JC\_Data/JC\_Edt/lnk/20161105194206236.pdf
- Bademci, H. Ö., Karadayı, E. F., & Vural, N. B. (2016). Exploring school exclusion through the perspective of child labourers living in Sultanbeyli, on the periphery of Istanbul, Turkey. *International Journal of Educational Development*, *50*, 51–58. https://doi.org/10.1016/j.ijedudev.2016.06.001
- Baird, M. (2010). Service Delivery in Fragile and Conflict-Affected States. Retrieved from https://openknowledge.worldbank.org/bitstream/handle/10986/9203/WDR2011\_0003.pdf?sequenc e=1&isAllowed=y
- Basmeh & Zeitooneh. (2017). Barriers to Education for Female and Male Syrian Youth in Shatila and Bourjal Barajneh. Retrieved from https://www.alnap.org/system/files/content/resource/files/main/bz\_november\_2017\_-\_barriers\_to\_education\_for\_female\_and\_male\_syrian\_youth\_in\_shatila\_and\_bourj\_al\_barajneh.pdf
- Basmeh & Zeitooneh. (2018). Non-Formal Education | Basmeh & Zeitooneh. Retrieved from http://www.basmeh-zeitooneh.org/our-programs/non-formal-education
- Batley, R., & Mcloughlin, C. (2010). Engagement with non-state service providers in fragile states: reconciling state-building and service delivery. *Development Policy Review*, 28(2), 131–154. Retrieved from http://onlinelibrary.wiley.com/doi/10.1111/j.1467-7679.2010.00478.x/full
- Berry, C. (2009). A Framework for Assessing the Effectiveness of the Delivery of Education Aid in Fragile States. *Journal of Education for International Development*. (4). Retrieved from http://toolkit.ineesite.org/toolkit/INEEcms/uploads/1023/A\_Framework\_Assessing\_Effectiveness.p df
- Blatter, J., & Haverland, M. (2012). Designing Case Studies: Explanatory Approaches in Small-N Research. Advance online publication. https://doi.org/10.1057/9781137016669

- Boustani, M., Carpi, E., Gebara, H., & Mourad, Y. (2016). Responding to the Syrian crisis in Lebanon: Collaboration between aid agencies and local governance structures. Retrieved from http://pubs.iied.org/pdfs/10799IIED.pdf
- Boustani, M., Carpi, E., Gebara, H., & Mourad, Y. (2017). Enhanced Local Coordination for Effective Aid Provision: The Case of Lebanon. Retrieved from http://website.aub.edu.lb/ifi/publications/Documents/policy\_memos/2016-2017/20170104 enhanced local.pdf
- Brinkerhoff, D. W. (2010). Developing Capacity in Fragile States. Public Administration and Development, 30(1), 66–78. https://doi.org/10.1002/pad.545
- Burde, D., Guven, O., Kelcey, J., Lahmann, H., & Al-Abbadi, K. (2015). What Works to Promote Children's Educational Access, Quality of Learning, and Wellbeing in Crisis-Affected Contexts: Education Rigorous Review. Retrieved from http://s3.amazonaws.com/ineeassets/resources/Education-emergencies-rigorous-review-2015-10.pdf
- Burde, D., Kapit, A., Wahl, R. L., Guven, O., & Skarpeteig, M. I. (2017). Education in Emergencies: A Review of Theory and Research. *Review of Educational Research*, 87(3), 619–658. https://doi.org/10.3102/0034654316671594
- Cherri, Z., Arcos González, P., & Castro Delgado, R. (2016). The Lebanese-Syrian crisis: impact of influx of Syrian refugees to an already weak state. *Risk Management and Healthcare Policy*, 9, 165–172. https://doi.org/10.2147/RMHP.S106068
- Concern Worldwide. (2018). Our Work in Lebanon. Retrieved from https://www.concern.net/where-we-work/middle-east/lebanon
- Danish Refugee Council. (2014). Strategic Program Document 2014DRC Lebanon. Retrieved from https://drc.ngo/media/1193992/DRC-Lebanon-Strategic-Program-Document-2014.pdf
- Deane, S. (2016). Syria's Lost Generation: Refugee Education Provision and Societal Security in an Ongoing Conflict Emergency. *IDS Bulletin*, 47(3). https://doi.org/10.19088/1968-2016.143
- Dionigi, F. (2016). The Syrian Refugee Crisis in Lebanon: State Fragility and Social Resilience.
- Echessa, E., & Pinnock, H. (2012). The lost learners: helping excluded children get an education in conflict-affected fragile states. *Compare: a Journal of Comparative and International Education*, 42(2, SI), 325–336. https://doi.org/10.1080/03057925.2012.652817
- El-Ghali, H. A., Ghalayini, N., & Ismail, G. (2016). Responding to Crisis: Syrian Refugee Education in Lebanon. Retrieved from http://website.aub.edu.lb/ifi/publications/Documents/policy\_memos/2015-2016/20160406 responding to crisis.pdf
- François, M., & Sud, I. (2006). Promoting Stability and Development in Fragile and Failed States. *Development Policy Review*, 24(2), 141–160. https://doi.org/10.1111/j.1467-7679.2006.00319.x

- Frederickson, G. H., Smith,, K. B., Larimer, C. W., & Licari, M. J. (2016). The Public Administration Theory Primer. Retrieved from https://rbb85.files.wordpress.com/2014/08/the public administration1-theory-primer.pdf
- The Fund for Peace. (2018). Fragile States Index. Country Dashboard Lebanon. Retrieved from http://fundforpeace.org/fsi/country-data/
- Government of Lebanon, & United Nations. Lebanon Crisis Response Plan 2015-2016.
- Government of Lebanon, & United Nations. (2014). Lebanon Crisis Response Plan 2015-2016. Retrieved from https://reliefweb.int/sites/reliefweb.int/files/resources/2015-2016\_Lebanon\_CRP\_EN.pdf
- Government of Lebanon, & United Nations. (2015). Lebanon Crisis Response Plan 2015-2016: Year Two. Retrieved from https://reliefweb.int/sites/reliefweb.int/files/resources/20151223\_LCRP\_ENG\_22Dec2015-

full%20version%29\_s.pdf

- Government of Lebanon, & United Nations. (2018). Lebanon Crisis Response Plan 2017-2020: 2018 Update. Retrieved from https://reliefweb.int/sites/reliefweb.int/files/resources/LCRP2018 EN Full 180122.pdf
- Hirschmann, K. (2016). *Wie Staaten schwach werden: Fragilität von Staaten als internationale Herausforderung*. Frankfurt/Main: Zarbock GmbH & Co. KG.
- Human Rights Watch. (2016). Growing up without an education: Barriers to Education for Syrian Refugee Children in Lebanon. Retrieved from https://www.hrw.org/sites/default/files/report\_pdf/lebanon0716web\_1.pdf
- INEE. (2010). Minimum Standards for Education: Preparedness, Response, Recovery. Retrieved from http://toolkit.ineesite.org/toolkit/INEEcms/uploads/1012/INEE\_Minimum\_Standards\_Handbook\_2 010(HSP)-English\_LoRes.pdf
- Inter-Agency Network for Education in Emergencies. (2009). Appropriate and Effective Financing Modalities and Channels for Education in Fragile Situations. Retrieved from http://s3.amazonaws.com/ineeassets/resources/Appropriate\_Effective\_financing\_Modalities\_Channels\_for\_Ed\_in\_Fragile\_Situ ations2.pdf
- Interesos. (2018). Lebanon | INTERSOS. Retrieved from https://www.intersos.org/en/what-wedo/lebanon/
- International Rescue Committee. (2017). Lebanon: Strategy Action Plan. Retrieved from https://www.rescue.org/sites/default/files/document/616/lebanonexternalsap-finaledit17alm.pdf
- International Rescue Committee, & Global Ties for Children at New York University. (2017a). An EIE Research-Practice Partnership: Learning to improve academic and social-emotional outcomes.

Retrieved from

https://static1.squarespace.com/static/583af1fb414fb5b3977b6f89/t/59bdb9f76f4ca3d8bb13c7ed/15 05606138131/13\_PromisingPractices\_\_IRC\_WEB.pdf

- International Rescue Committee, & Global Ties for Children at New York University. (2017b). IRC Healing Classrooms Retention Support Programming Improves Syrian Refugee Children's Learning In Lebanon: 3EA Impact Report 1.1. Retrieved from https://steinhardt.nyu.edu/scmsAdmin/media/users/mhm327/3ea\_docs/3EA\_-\_Impact\_report.pdf
- Janmyr, M. (2016). Precarity in Exile: The Legal Status of Syrian Refugees in Lebanon. *Refugee Survey Quarterly*. (35), 58–78. https://doi.org/10.1093/rsq/hdw016
- Jennings, Z. (2016). Impact of the provision of school lunch on attendance in remote rural Jamaican primary schools. *International Journal of Educational Development*, 46, 74–81. https://doi.org/10.1016/j.ijedudev.2015.09.006
- Justino, P. (2016). Supply and demand restrictions to education in conflict-affected countries: New research and future agendas. *International Journal of Educational Development*, 47, 76–85. https://doi.org/10.1016/j.ijedudev.2016.01.002
- Kagawa, F. (2005). Emergency Education: A Critical Review of the Field. *Comparative Education*, *41*(4), 487–503. https://doi.org/10.1080/03050060500317620
- Karam, F. J., Monaghan, C., & Yoder, P. J. (2017). 'The students do not know why they are here': education decision-making for Syrian refugees. *Globalisation, Societies and Education*, 15(4), 448–463. https://doi.org/10.1080/14767724.2016.1222895
- Karam Foundation. (2016). Displaced Generation: The Landscape of Refugee Education in Syria, Turkey, Jordan, and Lebanon. Retrieved from https://static1.squarespace.com/static/542de954e4b037619e869735/t/57e57bafc534a58cf4e8da7c/1 474657203399/Education+Report.pdf
- Kirk, J. (2007). Education and fragile states. *Globalisation, Societies and Education*, 5(2), 181–200. Retrieved from https://doi.org/10.1080/14767720701425776

Kisthardt, L. (2013). Confessionalism in Lebanon: Reflection and an analysis of issues. Retrieved from

https://collected.jcu.edu/cgi/viewcontent.cgi?referer=https://www.google.nl/&httpsredir=1&article =1014&context=honorspapers

- Langsten, R. (2017). School fee abolition and changes in education indicators. *International Journal of Educational Development*, *53*, 163–175. https://doi.org/10.1016/j.ijedudev.2016.12.004
- Makhzoumi Foundation. (2018). Makhzoumi Foundation: Our Programmes: Literacy with Crafts for Youth in Lebanon - AVSI. Retrieved from http://makhzoumi-foundation.org/vocationaltraining/#1492505291365-2f1af64e-da57

- Masterson, D. (2016). Giving Better: Lessons from Cash Grants for Syrian Refugees in Lebanon. Retrieved from http://website.aub.edu.lb/ifi/publications/Documents/policy\_memos/2015-2016/20160225\_impact\_cash\_assistance.pdf
- Matland, R. E. (1995). Synthesizing the Implementation Literature: The Ambiguity-Conflict Model of Policy Implementation. *Journal of Public Adminsitration Research and Theory*. (5). Retrieved from http://www.jstor.org/stable/1181674
- Maxwell, J. A. (2008). Designing a Qualitative Study, 214–253. Retrieved from https://in.sagepub.com/sites/default/files/upm-binaries/23772\_Ch7.pdf
- McLaughlin, C., & AL-Hroub, A. (2016). Understanding and Supporting Vulnerable Children in Schools of Lebanon. Retrieved from http://website.aub.edu.lb/ifi/publications/Documents/policy\_memos/2016-2017/20161011 vulnerable children.pdf
- Ministry of Education and Higher Education. (2014). Reaching All Children with Education in Lebanon R.A.C.E. Retrieved from http://www.mehe.gov.lb/uploads/file/2015/Feb2015/Projects/RACEfinalEnglish2.pdf
- Ministry of Education and Higher Education. (2016). Reaching All Children with Education: RACE II (2017-2021). Retrieved from http://www.mehe.gov.lb/uploads/file/2016/Oct/RACE%20II\_FINAL%20Narrative\_29AUG2016.p df
- Morais de Sa e Silva, M. (2015). Conditional cash transfers and improved education quality: A political search for the policy link. *International Journal of Educational Development*, *45*, 169–181. https://doi.org/10.1016/j.ijedudev.2015.09.003
- Norwegian Refugee Council. (2017). Non-formal education programming: An approach to increasing enrollment into the formal system. Retrieved from https://static1.squarespace.com/static/583af1fb414fb5b3977b6f89/t/59bdbb81cf81e06811307251/1 505606531272/10\_PromisingPractices\_NRC\_WEB.pdf
- Norwegian Refugee Council. (2018). NRC in Lebanon | NRC. Retrieved from https://www.nrc.no/countries/middle-east/Lebanon
- OECD. (2008). Service Delivery in Fragile Situations: Key Concepts, Findings and Lessons. *OECD Journal on Development*, 9(3), 7–60. https://doi.org/10.1787/journal\_dev-v9-art26-en
- Ontario Human Rights Commission. (2001). An Intesectional Approach to Discrimination: Addressing Multiple Grounds in Human Rights Claims. Retrieved from http://www.ohrc.on.ca/sites/default/files/attachments/An\_intersectional\_approach\_to\_discriminatio n%3A\_Addressing\_multiple\_grounds\_in\_human\_rights\_claims.pdf

- Oxfam. (2015). Lebanon: Looking ahead in times of crisis: Taking stock of the present to urgently build sustainable options for the future.
- Peace Security and Development Network. (2009). Working Paper I. The Fragile States discourse unveiled. Retrieved from https://ris.utwente.nl/ws/portalfiles/portal/5146692
- Premiere Urgence International. (2018). Mission Lebanon. Retrieved from https://www.premiereurgence.org/en/mission/lebanon/
- Rainey, H. G. (2014). Understanding and Managing Public Organizations (5th ed.): Jossey-Bass. A Wiley Brand.
- René Moawad Foundation. (2018). Supporting vulnerable girls and boys with learning, soft and hard skills and employability programmes in the North René Moawad Foundation. Retrieved from http://www.rmf.org.lb/project/supporting-vulnerable-girls-and-boys-with-learning-soft-and-hard-skills-and-employability-programmes-in-the-north/#!
- Rose, P. (2007a). NGO provision of basic education: alternative or complementary service delivery to support access to the excluded? *Compare: a Journal of Comparative and International Education*, 39(2), 219–233. https://doi.org/10.1080/03057920902750475
- Rose, P. (2007b). Supporting non-state providers in basic education service delivery. Retrieved from http://www.create-rpc.org/pdf\_documents/PTA4.pdf
- Rose, P., & Greeley, M. (2006). Education in fragile states: Capturing lessons and identifying good practice. DAC Fragile States Group. Retrieved from http://toolkit.ineesite.org/resources/ineecms/uploads/1096/Educ\_Fragile\_States\_Capturing\_Lessons .PDF
- Salti, N., & Chaaban, J. (2010). The Role of Sectarianism in the Allocation of Public Expenditure in Postwar Lebanon. *International Journal of Middle East Studies*, 42(04), 637–655. https://doi.org/10.1017/S0020743810000851
- Save the Children. (2009). Barriers to Accessing Primary Education in Conflict-Affected Fragile States. Retrieved from https://wcfia.harvard.edu/files/wcfia/files/2942.pdf
- Save the Children. (2015). What do Children Want in Times of Emergency and Crisis? They Want an Education. Retrieved from https://www.savethechildren.org/content/dam/global/reports/education-and-child-protection/what-children-want.pdf
- Save the Children Lebanon. (2018). Education | Lebanon | Save the Children. Retrieved from https://lebanon.savethechildren.net/what-we-do/education
- Search for Common Ground. (2015). Final Evaluation Report "Rainbow of Hope" Project Lebanon. Retrieved from https://www.sfcg.org/wp-content/uploads/2015/10/LIB\_EV\_July15\_Rainbow-of-Hope-project.pdf

- Seawright, J., & Gerring, J. (2008). Case Selection Techniques in Case Study Research: A Menu of Qualitative and Quantitative Options. *Political Research Quarterly*. (61), 294–308. Retrieved from http://www.jstor.org.ezproxy2.utwente.nl/stable/pdf/20299733.pdf?refreqid=excelsior:cb48d2a024 6b7a609808a848be5c0807
- Shuayb, M., Makkouk, N., & Tuttunji, S. (2014). Widening Access to Quality Education for Syrian Refugees: The Role of Private and NGO Sectors in Lebanon. Retrieved from http://www.lebanesestudies.com/wp-content/uploads/2014/09/Widening-Access-to-Quality-Education-for-Syrian-Refugees-FINAL.pdf

Sonbola. (2018). Programme Information. Retrieved from http://www.sonbola-ngo.com/programmes

- Stel, N., & van der Borgh, C. (2017). Political Parties and Minority Governance in Hybrid Political Orders: Reflections from Lebanon's Palestinian Settlements and Kosovo's Serbian Enclaves. *Journal of Intervention and Statebuilding*, 11(4), 490–510. https://doi.org/10.1080/17502977.2017.1376948
- Strategic Policy Impact and Research Unit. (2007). Education aid in fragile states: Delivering it effectively: ODI Strategic Policy Impact and Research Unit (SPIRU) Papers 1. Retrieved from https://www.odi.org/sites/odi.org.uk/files/odi-assets/publications-opinion-files/5048.pdf
- Sunny, B. S., Elze, M., Chihana, M., Gondwe, L., Crampin, A. C., Munkhondya, M.,. . . Glynn, J. R. (2017). Failing to progress or progressing to fail? Age-for-grade heterogeneity and grade repetition in primary schools in Karonga district, northern Malawi. *International Journal of Educational Development*, 52, 68–80. https://doi.org/10.1016/j.ijedudev.2016.10.004
- Turner, L. (2015). Explaining the (Non-)Encampment of Syrian Refugees: Security, Class and the Labour Market in Lebanon and Jordan. *Mediterranean Politics*, 20(3), 386–404. https://doi.org/10.1080/13629395.2015.1078125
- UNESCO, UNFPA, UNHCR, UNICEF, & Save the Children International. (2014). Situation Analysis of Youth in Lebanon Affected by the Syrian Crisis. Retrieved from https://reliefweb.int/sites/reliefweb.int/files/resources/YSA-SyriaCrisis-FullReport.pdf
- UNHCR, UNICEF, & United Nations World Food Programme. (2015). Vulnerability Assessment of Syrian Refugees in Lebanon 2015. Retrieved from https://documents.wfp.org/stellent/groups/public/documents/ena/wfp280798.pdf
- UNHCR, UNICEF, & United Nations World Food Programme. (2016). Vulnerability Assessment of Syrian Refugees in Lebanon 2016. Retrieved from https://reliefweb.int/sites/reliefweb.int/files/resources/VASyR2016.pdf
- UNHCR, UNICEF, & United Nations World Food Programme. (2017). Vulnerability Assessment of Syrian Refugees in Lebanon 2017. Retrieved from https://reliefweb.int/sites/reliefweb.int/files/resources/VASyR%202017.compressed.pdf

- UNICEF, UNHCR, & World Food Programme. (2017). Vulnerability Assessment of Syrian Refugees in Lebanon. VASYR 2017. Retrieved from https://data2.unhcr.org/en/documents/download/61312
- United Nations Children's Fund. (2011). The Role of Education in Peacebuilding: Case Study: Lebanon. Retrieved from
  - https://www.tandfonline.com/doi/pdf/10.1080/14767724.2013.858988?needAccess=true
- United Nations Children's Fund Lebanon. (2016). Lebanon Country Office Education Strategy Note. Retrieved from
  - http://files.unicef.org/transparency/documents/Lebanon\_Education\_StrategyNote\_3MAY2016.pdf
- United Nations Development Programme. (2009). Human Development Report 2009: Overcoming barriers: Human mobility and development. Human development report: Vol. 2009. Houndmills: Palgrave Macmillan. Retrieved from

http://hdr.undp.org/sites/default/files/reports/269/hdr\_2009\_en\_complete.pdf

- United Nations Educational, Scientific and Cultural Organization. (2017). *Education 2030: Ensure inclusive and equitable quality education and promote lifelong learning opportunities for all.* Retrieved from http://unesdoc.unesco.org/images/0024/002456/245656E.pdf
- United Nations High Commissioner for Refugees. (2017). The Ministry of Education and Higher Education and UN Agencies Launch This Year's Back to School Campaign Celebrating School Heroes. Retrieved from

https://reliefweb.int/sites/reliefweb.int/files/resources/EN\_2017%20UNICEF%20Back%20to%20S chool%20Camapaign%20aunch-%20PRL.pdf

- United Nations High Commissioner for Refugees, & United Nations Development Programme.
  (2017). 3RP. Regional Refugee and Resilience Plan 2017-2018: In Response to the Syria Crisis.
  2017 Progress Report. Retrieved from http://www.3rpsyriacrisis.org/wp-content/uploads/2017/10/3RP-Progress-Report-17102017-final.pdf
- United Nations High Commissioner for Refugees, & United Nations Development Programme.
   (2018). 3RP. Regional Refugee and Resilience Plan 2018-2019: In Response to the Syria Crisis.
   Regional Strategic Overview. Retrieved from https://reliefweb.int/sites/reliefweb.int/files/resources/3RP-Regional-Strategic-Overview-2018-

19.pdf

United Nations High Commissioner for Refugees Lebanon. (2016). Back to School Campaign 2015-2016. Retrieved from

https://reliefweb.int/sites/reliefweb.int/files/resources/Backtoschoolbrochure2016portal.pdf

United Nations High Commissioner for Refugees Lebanon. (2018). Partners for project implementation - UNHCR Lebanon. Retrieved from http://www.unhcr.org/lb/project-partnerships

- United Nations System in Lebanon. (2017). United Nations Strategic Framework (UNSF) Lebanon 2017-2020. Retrieved from https://reliefweb.int/sites/reliefweb.int/files/resources/UNSF%20Lebanon%202017-2020-034537.pdf
- United States Agency for International Development. (2018). IDEA Country Dashboard Lebanon. Retrieved from https://idea.usaid.gov/cd/lebanon?comparisonGroup=region#print-view
- United States Department of State. (2017). Evaluation Report: Evaluating the Effectiveness of Shelter, Health, and Education Programs for Irawi and Syrian Refugees in Lebanon. Retrieved from https://www.state.gov/documents/organization/278594.pdf
- UTOPIA. (2018). Child Development Programme. Retrieved from http://utopialebanon.org/category/child-development/
- Visconti, L., & Gal, D. (2018a). Regional collaboration to strengthen education for nationals & Syrian refugees in Arabic speaking host countries. *International Journal of Educational Development*, 61, 106–116. https://doi.org/10.1016/j.ijedudev.2017.10.022
- Visconti, L., & Gal, D. (2018b). Regional collaboration to strengthen education for nationals & Syrian refugees in Arabic speaking host countries. *International Journal of Educational Development*, 61, 106–116. https://doi.org/10.1016/j.ijedudev.2017.10.022
- Wofford, M. C., & Tibi, S. (2018). A human right to literacy education: Implications for serving Syrian refugee children. *International Journal of Speech-Language Pathology*, 20(1), 182–190. https://doi.org/10.1080/17549507.2017.1397746
- The World Bank Group Middle East and North Africa Region. (2015). Lebanon: Promoting Poverty Reduction and Shared Prosperity: A Systematic Country Diagnostic. Retrieved from http://documents.worldbank.org/curated/en/698161468179049613/pdf/97498-CAS-P151430-SecM2015-0202-IFC-SecM2015-0073-MIGA-SecM2015-0047-Box391476B-OUO-9.pdf

World Vision. (2017). Country Information Lebanon. Retrieved from https://www.wvi.org/lebanon

### 7. Data Appendix

## 7.1 Historical Background of Lebanon: Conflict and Sectarian Tensions as Fragility Drivers

"The Syria crisis in Lebanon is more than a refugee crisis; it is a refugee crisis compounded by an already fragile socioeconomic and political context and by a constrained public system, both in terms of resources and capacity" (MEHE, 2014, p. 3). This means that the challenges of the refugee crisis in Lebanon cannot be viewed in isolation but embedded in the context of the historical development. Therefore, this section elaborates on Lebanon's history in terms of Lebanon's relations to Syria and in terms of the predominant confessionalism that impacts on today's political-administrative system and thus education delivery.

Lebanon and Syria are tied together with a shared history. Both territories were part of the Ottoman empire from the 16<sup>th</sup> century until 1915, when Western powers divided the Ottoman territories during World War I (Kisthardt, 2013). During the Ottoman empire, sectarianism as "deployment of religious heritage as a primary marker of modern political identity" emerged with the establishment of confessional councils to oversee the self-government in Lebanon's diverse religious communities (United Nations Children's Fund [UNICEF], 2011, p. 19). This system encompassed autonomy over the educations sector and every religious community had its own educational sponsors. With the founding of the Mutasarrifate of Mount Lebanon in 1860, greater Lebanon (including nowadays Syria) was divided into six administrative districts along sectarian lines, manifesting a sectarian consciousness in the public identity (UNICEF, 2011). At that time, also ties between the sectarian communities and foreign powers developed, influencing the education sector. Maronite Christian communities were affiliated with France, Protestant Christian communities with Britain and the United States, and Russians, Germans, Greeks and other Europeans also sought influence in the region. They spread "education and language along sectarian lines, with the ruling Ottomans developing schools for Sunni Muslims" (UNICEF, 2011, p. 20). Arabic was ascribed a literary function in schools and foreign languages were assigned scientific functions in education. Furthermore, national Christian and Islamic schools developed as rivals of foreign schools, laying the foundation for private and parochial schooling (UNICEF, 2011). Both, the multilinguistic education delivery and the emergences of private education, impact on the Lebanese education sector today.

Following the Sykes-Picot Agreement, the French Mandate from 1920 to 1943 covered the territories of Syria and Lebanon. The French introduced their language in the education sector and consolidated sectarianism by privileging Maronite elites, granting them vast political powers in the parliamentary structure (Kisthardt, 2013). They conducted the last official census in 1932, which formed the basis for the sect-based power-sharing formula represented in the National Pact of 1943 (UNICEF, 2011). The National Pact was an unwritten agreement between Christian and Muslim leaders who

declared Lebanon a neutral sovereign country with Arab character, which does not seek unity with Syria and is independent from France. More importantly, it declared a 6:5 ratio of Christian to Muslim representation in the parliament and attributed the position of the President to a Maronite Christian, the position of the Prime Minister to a Sunni Muslim and the position of the Speaker of Parliament to a Shia Muslim, thus institutionalising the consociationalism along confessional lines known today (Kisthardt, 2013). Since then, the demographic composition has changed towards the Muslim community, but no other consensus has been conducted to not disrupt the delicate balance of power. In the years after Lebanon's independence, the power-imbalances following the underrepresentation of Muslims resulted in the inequitable development in areas where marginalized groups such as Shia lived and set stage to many violent conflicts.

In 1967, the Arab-Israeli War lead to a displacement of Palestinians who sought refuge in Lebanon. It also led to a flow of arms from Syria to the Palestine Liberation Organization which operated in Palestinian refugee camps inside Lebanon, thus destabilising the country and upsetting the power balance due to the influx of Muslims (UNICEF, 2011).

In 1975, the 15 year-long Civil War between Christians, Muslims and Palestinians evolved, in which around 150 000 people were killed until the end in 1990 (UNICEF, 2011). The injustice of the confessional system, the Palestinians in Lebanon as well as regional and international actors following their interests are blamed for the protracted nature of the civil war (UNICEF, 2011). During that time, the conflict and the complex composition of armed groups left the political system paralysed and impacted on the state capacity. The state failure compromised public schooling and lead to the spread of private and parochial schooling which reproduced the sectarian divisions (UNICEF, 2011). The end of the Civil War was declared in 1989 with the Tarif Accord, the Document of National Reconciliation. It defined Lebanon as a parliamentary democracy with a free economy favouring privatization as well as an Arab identity, and itcalled for a new power sharing agreement for the Parliament that divides the power between Christians and Muslims more equally (Kisthardt, 2013). However, the new formula of sectarian balance did not address structural shortcomings of such a system. The Tarif agreement also resulted in an education reform that encouraged standardized curricula, the protection of private schooling and the development of public schooling. Furthermore, it aimed at strengthening a national Lebanese identity through common education components in language and history (UNICEF, 2011).

After the Civil War, tensions and grievances and the polarization of the society remained. In 2000, Israel withdrew its occupation from the Lebanese areas in the South and the Bekaa valley after 22 years of conflict (UNICEF, 2011). From 2004 to 2008, several political assassinations and other violent acts took place. The most prominent one was the assassination of the former Prime Minister Hariri in 2005, which caused the peaceful Cedar Revolution as well as enormous protests, the resignation of the government and the withdrawal of the Syrian army, which was present in Lebanon since 1976 (UNICEF,

2011). Furthermore, the July War in 2006 between Israel and the Lebanese Hezbollah destructed a lot of Lebanon's educational infrastructure, especially in the South and the Bekaa valley (UNICEF, 2011).

Between 2006 and 2007, the protests created the March 8 and March 14 movements, which constitute political parties today. The March 14 alliance was against the Syrian influence in the Lebanese government and lead to the removal of Syria's forces from Lebanon. The Hezbollah associated March 8 alliance was in favour of the Syrian regime and demanded the Lebanese government to resign. This paralyzed Lebanon's political system once again and raised awareness for sectarian politics, resulting in street clashes (UNICEF, 2011). In 2007, clashes between the Al-Qaeda-inspired Fateh al-Islam and the Lebanese army lead to the Nahr el-Bared Crisis, a Palestinian refugee camp in and around which a three-month war took place and raised tensions between the communities (UNICEF, 2011). In 2008 and 2011 tensions between the Hezbollah and the government as well as between Alawi Muslims (pro March 8) and Sunni Muslims (pro March 14) erupted, showing the sectarian divisions. Since the beginning of the Syrian conflict in 2011, opposing views on supporting or resisting the Syrian regime drew people to the streets.

It is in the light of this history of conflict and sectarian tension that the situation of Lebanon today needs to be viewed in order to understand Lebanon's stance on refugees today. Referring back to the theoretical framework on state fragility, the categorization of origins of fragility by Hirschmann (2016) proves to be a fitting concept. Taking into account the fours situations out of which fragility develops, the case of Lebanon can be traced back to the fourth category of a constructed state as a consequence of colonial rule. The division of the region into spheres of British and French influence following the Sykes-Picot agreement and Lebanon's occupation by France lead to an artificial construction of state borders which did not respect the affiliations of the residing communities. The Christian Maronites, Shia and Sunni Muslims, Druze and other ethnic communities were drawn together into a state construct which repeatedly lead to conflict and clashes along sectarian lines due to socioeconomic inequalities. The recurrent eruptions of violence are a symptom of this lack of national identity. Without the colonial oppression, these ethnic communities would not have merged into a common state. Therefore, the sense of belonging is transferred to the sub-state level which can be seen in the sectarian organization of public life, which decreases the legitimacy of the state structure as people do not feel connected to it (Hirschmann, 2016).

The historical background also shows how the sectarian divide lead to recurrent eruptions of conflict that hamper the capacity for action of the government and how regional and international powers play on these divisions to follow their own interests in Lebanon. This is illustrated with the education sector on which sectarianism and the involvement of international actors impacted. Education has been provided to the own community but the responsibility for other ethnic communities was undermined by the ethnic appropriation of education. Characteristics of the education sector such as the multilinguistic

language of instruction and the flourishing private sector, which will be more elaborated on in the section on the Lebanese education sector today, date far back in time. Education has been subject to politicization and instrumentalization, first to incite the ethnic communities, playing on their unequal treatment in the country, and after the civil war to propagate national unity through the curriculum. Moreover, the historical background presents how the Palestinians seeking refuge in Lebanon jeopardized the sectarian equilibrium which is relevant for understanding the reaction to the Syrian refugees today.

## 7.2 Operationalization Table

Theoretical concepts	Variables	Conceptualization of the Variables	Operationalization of the Variables
Barriers to education	Demand side barriers	Cost of education (direct and indirect)	Direct: Abolition/waiving of tuition Indirect: Presence of school feeding programme Provision of stationary/books Provision of transport Provision/no school uniforms
		Coping mechanisms related to costs: Child labour Coping mechanisms related to costs and insecurity: Child marriage/supply- side barrier insecurity	Cash transfer Programmes Interventions that alleviate family poverty Restructuring schools to reconcile work with education Awareness session on importance to education Involvement of family and community High share of female teachers
		Psychological Stress and Trauma	Access to psychological counselling Access to psycho-social activities
		Over-age for Grade Level	Repetition rates
	Supply side barriers	Lack of Language Skills/Curriculum	Language support Multilingual teaching material Literacy and numeracy support Homework support Remedial support
		Lack of Documentation	Easing enrolment
		Discrimination and Violence	Easing exam registration Non-violent teaching practices Presence of Inclusive classroom approach Absence of bullying Absence of (corporal) punishment Absence of politicised curriculum
Effectiveness	Monitoring Instruments	Monitoring instruments present/not	Assessments at student level Assessments at teacher level Assessments at administrative/organizational level
	Goal Achievement	Goal achievement/not	Project outputs

<sup>iv</sup> Due to prolonged validation processes the evaluations of the education sector (in comparison to e.g. WASH) were not completed but initiated in 2017 (United Nations Children's Fund [UNICEF], 2016). The researcher could therefore not draw on them.

<sup>&</sup>lt;sup>i</sup> Madrasahs are Islamic religious seminaries (Rose, 2007b).

<sup>&</sup>lt;sup>iii</sup> Among the projects that were contacted are the *Quality Instruction Towards Access and Basic Education Improvement* project by MEHE, World Learning, Ana Aqra and AMIDEAST, the *Non-Formal Education* and *Peace Education* project by the NGO Basmeh & Zeitooneh, the *Sonbola Taleem* and *Sonbola Tamkeen* programme by the NGO Sonbola, and the education approach of the NGO Tahaddi.